THERMAL MANAGEMENT OF THREE-DIMENSIONAL INTEGRATED CIRCUITS USING INTER-LAYER LIQUID COOLING

A Doctoral Dissertation Presented to The Academic Faculty

by

Calvin R. King, Jr.

In Partial Fulfillment of the Requirements for the Degree of Doctor of Philosophy in Electrical and Computer Engineering in the School of Electrical and Computer Engineering



Georgia Institute of Technology August 2012

THERMAL MANAGEMENT OF THREE-DIMENSIONAL INTEGRATED CIRCUITS USING INTER-LAYER LIQUID COOLING

Approved by:

Dr. James Meindl, Advisor School of Electrical and Computer Engineering *Georgia Institute of Technology*

Dr. Gary May School of Electrical and Computer Engineering *Georgia Institute of Technology*

Dr. Sung-kyu Lim School of Electrical and Computer Engineering *Georgia Institute of Technology* Dr. Muhannad Bakir, Co-advisor School of Electrical and Computer Engineering *Georgia Institute of Technology*

Dr. Oliver Brand School of Electrical and Computer Engineering *Georgia Institute of Technology*

Dr. Samuel Graham School of Mechanical Engineering *Georgia Institute of Technology*

Date Approved: [May 04, 2012]

Dedicated to my wife, Mya, my two sons, Calvin III and Jackson, my parents, Calvin Sr. and Sarah, and my sister and brother, Kimberly and Kyle for their encouragement, love, and support.

ACKNOWLEDGEMENTS

"What we are is God's gift to us. What we become is our gift to God." I would like to thank God for giving me the ability and endurance to complete three graduate degrees at Georgia Institute of Technology. I intend to use the education that I been blessed to receive as a tool to enable me to fulfill the purpose that God has for my life and to make a positive contribution to society.

I would like to give special thanks to my research advisor, Dr. James Meindl, for accepting me as a student in the Gigascale Integration Group. Dr. Meindl's insight and wisdom has greatly impacted the direction of my research. The meaningful discussions that we have shared during my graduate tenure have been truly inspirational to me.

My co-advisor, Dr. Muhannad Bakir, has had a monumental impact on my life. From training me in the cleanroom, assisting me in technical writing, and giving me advice, Dr. Bakir has played a key role in my personal and professional development.

I would like to thank my PhD dissertation committee members Dr. Gary May, Dr. Oliver Brand, Dr. Sung-Kyu Lim, and Dr. Samuel Graham for giving me suggestions and feedback that helped me to improve my research. My fellowship sponsors, IBM, SRC, Intel, Ford Foundation, NASA, Applied Materials, and FACES are greatly appreciated for the financial support that has enabled me to pursue my graduate studies.

To some of my closest friends, Delano Billingsley, Kevin Fairbanks, Quenton Strickland, Isaac Addae, Ayodeji Olojo, and Jermaine Smith, I appreciate all of the encouragement that you have given me over the years. I must also thank Ashley Johnson, Nicholas Brown, and Shamas Ummer for the many hours that we spent together studying for classes and qualifying exams.

With respect to helping me to build my academic foundation and confidence to pursue a Ph.D., I would like to thank Dr. Decatur Rogers, Dr. Didar Sohi, Michael Rodriguez, and Garland Wayne Williams. The MiRC cleanroom staff members deserve a big thank you for training me on the tools that I needed to complete the experimental component of my research, and I would like to take the ECE Graduate Affairs Office staff members for helping me to navigate through the non-research related issues of graduate school.

I would like to thank Jennifer Tatham Root and Tina Mosley for always assisting me and helping me to have a pleasant experience at Georgia Tech. Thank you to the past and present members of the GSI Group (Oluwafemi Ogunsola, Hiren Thacker, Bing Dang, Deepak Sekar, Farhana Zaman, and Ramasamy Ravindran) and to past and present members of the Integrated 3D Systems Group members (Jesal Zaveri, Yue Zhang, and Ashish Dembla) for all of the meaningful conversations, research collaborations, and enjoyable experiences.

Last but definitely not least, I would like to thank my wife Mya King, my sons Calvin King, III and Jackson King, my parents Dr. Calvin King, Sr. and Sarah King, my sister Kimberly King Davis and my brother Kyle King for their never-ending love, support, and encouragement throughout this journey.

TABLE OF CONTENTS

	Page
ACKNOWLEDGEMENTS	
LIST OF TABLES	vii
LIST OF FIGURES	
SUMMARY	xvii
1. INTRODUCTION AND BACKGROUND	1
	2
1.1. Thermal Management Challenges in High-performance ICs	2
1.1.1. Increasing Power Density	2
1.1.2. Leakage Power	3
1.1.3. Increasing Heat Sink Size	4
1.1.4. Heat Sink Thermal Resistance	6
1.1.5. Microprocessor Clock Frequency	7
1.2. Three Dimensional Integrated Circuits	8
1.3. Thermal Challenges in Three Dimensional Integrated Circuits	9
1.3.1. Challenges of Conventional Air-cooling Heat Sinks	10
1.3.2. Advanced Cooling of High-performance 3D Integrated Circuits	14
1.4. Summary of Research Objectives	16
1.5. Dissertation Outline	16

2. DEVELOPMENT OF AN INTER-LAYER LIQUID COOLING

PLATFORM FOR THERMAL MANAGEMENT OF 3D

INTEGRATED CIRCUITS

	2.1.	Development of an Inter-layer Liquid Cooling Platform for Thermal	
		Management of 3D Integrated Circuits	19
	2.2.	3D Inter-layer Liquid Cooling Platform Process Development and	
		Integration of Microchannel Heat Sink, Electrical TSVs, Fluidic	
		TSVs, Electrical I/Os, and Fluidic I/Os	21
	2.3.	Methods for Microchannel Heat Sink Capping	25
		2.3.1. Silicon-to-Silicon Bonding	26
		2.3.2. Polymer Overcoat	30
	2.4.	Microchannel Heat Sink Design	31
		2.4.1. Microchannel Heat Sink Theory	31
		2.4.2. Microchannel Heat Geometry and Thermal Resistance	36
		2.4.3. Microchannel Heat Geometry and Pressure Drop	38
	2.5.	Microchannel Heat Sink and Electrical TSV Co-design	43
		2.5.1. Microchannel Heat Geometry and Electrical TSV Density	44
		2.5.2. Microchannel Heat Geometry and Electrical TSV Performance	51
	2.6.	Summary	55
3.	DES	IGN AND FABRICATION OF ELECTRICAL AND FLUIDIC I/O	
	INTI	ERCONNECTS	57
	3.1.	Integrated Electrical and Fluidic I/O Technologies for 3D Inter-layer	
		Liquid Cooling	57
	3.2.	C4 Fluidic I/O Interconnects	59
	3.3.	Integrated C4 Electrical and Fluidic I/O Interconnects	63
		3.3.1. Integrated C4 Electrical and Rectangular Fluidic I/O	

		Interconnects	66
	3.4.	Air-gap C4 Fluidic I/O Interconnects	69
	3.5.	Integrated Polymer Sockets, Fluidic TSVs, Electrical I/Os and	
		Polymer Pipe Fluidic I/Os	71
	3.6.	Summary	75
4.	3D (CHIP ASSEMBLY USING ELECTRICAL AND FLUIDIC I/O	
	INT	ERCONNECTS	76
	4.1.	3D Chip Stacking Using Electrical and Fluidic I/O Interconnects	76
	4.2.	Assembly of C4 Electrical and C4 Fluidic I/Os	76
	4.3.	Assembly of C4 Electrical and Air-gap C4 Fluidic I/Os	80
	4.4.	Fluidic Testing of C4 Fluidic I/O Interconnects	86
	4.5.	Assembly Using Polymer Sockets Polymer Pipe Fluidic I/Os,	
		and Electrical I/Os	87
	4.6.	Summary	92
5.	THE	RMAL AND FLUIDIC TESTING OF THE 3D INTER-LAYER	
	LIQ	UID COOLING PLATFORM	93
	5.1.	Experimental Thermal Measurement for 3D Liquid Cooling	93
	5.2.	Fabrication of Top Layer in 3D Chip Stack	94
	5.3.	Fabrication of Bottom Layer in 3D Chip Stack	96
	5.4.	Fluidic Connectivity within the 3D Chip Stack	99
	5.5.	Platinum Heater / Thermometer Design and Fabrication	101
	5.6.	Electrical Connectivity in the 3D Chip Stack	104
	5.7.	Experimental Thermal Measurement Comparison of an Air-cooling	

		Heat Sir	nk vs. a Microchannel Heat Sink	105
		5.7.1.	Experimental Thermal and Fluidic Testing Setup	105
		5.7.2.	Heat Sink Volume Comparison for Air-cooling vs.	
			Microchannel Liquid Cooling	108
		5.7.3.	Thermal Resistance Analyssis for Air-cooling Heat Sink vs.	
			Microchannel Heat Sink	110
	5.8.	Hydraul	ic Analysis of Microchannel Heat Sink	112
	5.9.	Microflu	uidic Pin-fin Heat Sink	114
	5.10.	Summar	ry	117
6.	CO	NCLUSI	ON AND FUTURE WORK	118
	6.1.	Opportu	inities for Future Work	118
	6.2.	Conclus	sion	120
Ał	PPENI	DIX A: 1	PUBLICATIONS	124
RI	REFERENCES		127	
VI	/ITA		135	

LIST OF TABLES

		Page
Table 1.1:	ITRS projections for high-performance chip junction temperature	
	and power dissipation.	5
Table 1.2:	ITRS projections for the maximum number of high-performance	
	dice in a 3D chip stack and in a 3D package	11
Table 2.1:	ITRS data for high-performance microprocessors.	34
Table 2.2:	Summary of simulation results for thermal resistance as a function	
	of channel height (channel width = $100\mu m$).	38
Table 2.3:	Summary of simulation results for thermal resistance as a function	
	of channel height (channel width = $50\mu m$).	38
Table 2.4:	Summary of simulation results for thermal resistance as a function	
	of channel height (channel width = $150 \mu m$).	38
Table 2.5:	Summary of simulation results for pressure drop as a function of	
	channel height (channel width = $100\mu m$).	40
Table 2.6:	Summary of simulation results for pressure drop as a function of	
	channel width (channel height = $350 \mu m$).	41
Table 2.7:	Summary of simulation results for thermal resistance as a function	
	of pressure drop.	41
Table 2.8:	Microchannel heat sink dimensions for proposed 3D cooling scheme.	43
Table 2.9:	ITRS projections for high-performance chips: number of electrical	
	TSVs, minimum TSV pitch, maximum TSV aspect ratio, and	
	minimum die thickness	44
Table 2.10	: Electrical TSV density as a function of channel fin width and	

electrical TSV diameter.	50
Table 2.11: Electrical TSV resistance as a function of wafer thickness and	
TSV diameter.	52
Table 2.12: Electrical TSV capacitance as a function of wafer thickness and	
TSV diameter.	55
Table 3.1: Comparison of fluidic I/O interconnect technologies.	59
Table 3.2: Characterization of feature height of square-like fluidic I/O.	63
Table 3.3: Characterization of circular fluidic I/O.	63
Table 3.4: Feature sizes of various bump-pipe combinations before and after	
solder reflow.	66
Table 3.5: Feature sizes of various bump-pipe combinations before and after	
solder reflow.	68
Table 3.6: Summary of pressure drop calculations for fluidic I/O designs.	70
Table 3.7: Summary of chip area occupied by fluidic I/Os.	70
Table 4.1: Before and after reflow feature height parameters for assembly.	79
Table 4.2: Bonding process parameters.	80
Table 4.3: Bonding process parameters.	89
Table 5.1: Thin-film platinum heater characterization.	104
Table 5.2: Heat sink volume comparison between the Intel i7 air-cooling	
heat sink and the microchannel heat sink.	109
Table 5.3: Comparison of heat sink thermal resistance and chip average	
junction temperature for air cooling vs. microchannel liquid cooling.	113
Table 5.4: Comparison of microchannel and micro pin-fin heat sinks.	116

LIST OF FIGURES

		Page
Figure 1.1:	(a.) The first integrated circuit. (b.) Intel's Itanium Poulson	
	microprocessor containing over 3.1 billion transistors.	1
Figure 1.2:	Moore's Law Data.	2
Figure 1.3:	Historical power density of various commercial microprocessors.	3
Figure 1.4:	Chip power consumption as a function of temperature; leakage power	
	dependence on chip temperature	4
Figure 1.5:	Graph of increasing heat sink volume and heat sink fan power for	
	various Intel microprocessors	5
Figure 1.6:	Pentium Overdrive "chip and heat sink" from 1993 (bottom left);	
	standard Pentium 4 "heat sink only" from 2005	5
Figure 1.7:	Schematic of thermal resistances associated with a conventional	
	air-cooling heat sink.	6
Figure 1.8:	Figure 1.8: Frequency vs. Year for various Intel microprocessors.	7
Figure 1.9:	Examples of 3D integration using electrical TSVs (left) and	
	wire bonding (right).	9
Figure 1.10	: ITRS projected power dissipation for high performance and cost	
	performance microprocessors.	11
Figure 1.11	A schematic illustrating how heat flux increases from 2D to 3D	
	circuits.	12
Figure 1.12	Schematic of thermal resistances associated with a conventional	
	air-cooling heat sink.	13
Figure 1.13	: 3D thermal management using conventional air cooling.	13

Figure 1.14	: Heat transfer coefficient of various thermal buses.	15
Figure 2.1:	3D IC (a) processor-memory and (b) processor-processor thermal	
	management approaches using a microchannel heat sink inter-layer	
	liquid cooling.	20
Figure 2.2:	Schematic of chip-scale microchannel heat sink cooling scheme	
	for 3D integrated circuits.	22
Figure 2.3:	Schematic of wafer-level integration of microchannels, fluidic	
	through silicon vias, electrical through-silicon vias, and electrical	
	and fluidic I/Os to enable liquid cooling of high-performance 3D	
	integrated systems.	23
Figure 2.4:	Cross-sectional optical image and schematic of a polymer-capped	
	microchannel heat sink and integrated electrical through-silicon vias.	23
Figure 2.5:	(a) X-ray image and (b) cross-sectional optical image of a	
	silicon-capped microchannel heat sink and integrated electrical	
	through-silicon vias.	24
Figure 2.6:	SEM image of integrated electrical and fluidic I/O interconnects.	24
Figure 2.7:	SEM images of (left) microchannel heat sink and (right) integrated	
	thin-film platinum resistors and electrical I/Os.	25
Figure 2.8:	Top view of microchannel heat sink and fluidic TSVs after silicon	
	etching.	26
Figure 2.9:	Chemical reactions during the hydrophilic bonding process.	27
Figure 2.10	: Schematic of process flow for capping microchannels via	
	silicon-to-silicon bonding.	29
Figure 2.11	: A Schematic of process flow for capping microchannels via silicon-to	
	silicon bonding.	30

Figure 2.12: (a)Microchannel trenches are etched into the back side of the	
wafer. (b)Spin coat and polish Unity sacrificial polymer. (c) Spin	
coat and pattern Avatrel polymer sockets as an overcoat layer.	
(d) Simultaneous curing of Avatrel polymer and thermally	
decomposition of sacrificial polymer.	31
Figure 2.13: Schematic illustration of a microchannel heat sink concept	
demonstrated by Tuckerman and Pease [1.39].	32
Figure 2.14: Hydro-dynamic equations for microchannel heat sink theory.	35
Figure 2.15: Schematic illustration of a microchannel heat sink.	36
Figure 2.16: Simulation results for thermal resistance as a function of channel	
height for various fixed channel fin widths.	37
Figure 2.17: Simulation results for thermal resistance as a function of channel	
height (channel width = $150\mu m$).	39
Figure 2.18: Simulation results for pressure drop as a function of channel width	
(channel height = $350 \mu m$).	40
Figure 2.19: Simulation results for thermal resistance as a function of pressure	
drop.	42
Figure 2.20: Schematic illustration of proposed microchannel heat sink design.	43
Figure 2.21: ITRS projections for number of electrical TSVs per high-	
performance chip.	45
Figure 2.22: Schematic illustration of a wafer with integrated electrical TSVs	
and a microchannel heat sink.	46
Figure 2.23: Schematic of microchannel fin with integrated electrical TSVs.	47
Figure 2.24: Electrical TSV density as a function of channel fin width and	
electrical TSV diameter (20um).	48

Figure 2.25:	Electrical TSV density as a function of channel fin width and	
	electrical TSV diameter (5um).	49
Figure 2.26:	Electrical TSV resistance as a function of wafer thickness and TSV	
	diameter.	52
Figure 2.27:	Top view of an electrical TSV in silicon and the capacitances	
	associated with the TSV structure.	53
Figure 2.28:	Electrical TSV oxide capacitance as a function of wafer thickness	
	and TSV diameter.	54
Figure 3.1:	C4 fluidic I/O, C4 air-gap fluidic I/O, and polymer pipe fluidic I/O.	58
Figure 3.2:	Schematic of C4 electrical and fluidic I/O fabrication.	60
Figure 3.3:	SEM images of 30µm tall C4 fluidic I/Os (a.) before and (b) after	
	reflow.	61
Figure 3.4:	Optical images of C4 fluidic I/Os.	61
Figure 3.5:	(a.) C4 Fluidic I/O height uniformity before and after reflow; (b.)	
	C4 Fluidic I/O inner diameter uniformity before and after reflow.	62
Figure 3.6:	Feature Height Uniformity Shape Dependence – (a) circular	
	solder fluidic I/O and (b) square-like solder fluidic I/O.	63
Figure 3.7:	Schematic of the chip layout design for C4 electrical and fluidic I/Os.	64
Figure 3.8:	SEM image of integrated C4 electrical and fluidic I/Os after reflow.	64
Figure 3.9:	SEM images of integrated C4 electrical and rectangular C4 fluidic	
	I/Os.	66
Figure 3.10:	50µm tall, over-plated air-gap C4 fluidic I/Os (a) before and	
	(b) after reflow.	70
Figure 3.11:	X-ray image of air-gap C4 fluidic I/Os after solder reflow	71
Figure 3.12:	Schematic of wafer-level integration of polymer sockets, electrical	

	I/O and fluidic I/O interconnects for a 3D flip-chip package.	73
Figure 3.13:	SEM image of 15µm tall, 270µm and 60µm diameter polymer sockets.	74
Figure 3.14:	SEM image of 100µm diameter through-wafer fluidic interconnects.	74
Figure 3.15:	SEM image of 50µm tall solder bumps used as electrical I/O	
	Interconnects.	74
Figure 3.16:	SEM image of 60µm tall polymer pipes used as I/O fluidic	
	interconnects.	74
Figure 3.17:	Schematic showing how polymer fluidic I/Os can be used to route	
	liquid over longer distances in the 3D chip stack.	74
Figure 4.1:	Characterization and assembly of C4 fluidic I/Os.	77
Figure 4.2:	Schematic of assembly of C4 electrical and fluidic I/Os.	78
Figure 4.3:	SEM image of substrate with patterned copper rings (360µm outer	
	diameter) and integrated fluidic TSVs (left) and an SEM image	
	integrated copper pads and copper rings.	78
Figure 4.4:	Photograph of Finetech Fineplacer Lambda flip-chip bonder.	79
Figure 4.5:	Bonding process parameters.	80
Figure 4.6:	Characterization and assembly of air-gap C4 fluidic I/Os.	81
Figure 4.7:	3D assembly using air-gap C4 fluidic I/Os and electrical I/Os.	81
Figure 4.8:	X-ray image of air-gap C4 fluidic I/Os after assembly.	82
Figure 4.9:	X-ray image of air-gap C4 fluidic I/Os after assembly.	82
Figure 4.10:	SEM image of a 3D chip stack assembled using C4 fluidic	
	and electrical I/Os	83
Figure 4.11:	Optical image showing fluidic TSV clogged with solder.	84
Figure 4.12:	Cross-sectional optical image of a 3D chip stack assembled using	
	C4 fluidic and electrical I/Os.	84

Figure 4.13	: (a.) Integrated C4 electrical and rectangular C4 fluidic I/Os.	
	(b.) Copper pads fabricated on the substrate to facilitate assembly.	85
Figure 4.14	: X-ray images of successfully bonded 3D stack of chips with C4	
	electrical and rectangular fluidic I/Os.	86
Figure 4.15	: Experimental setup for fluidic testing of C4 fluidic I/Os.	87
Figure 4.16	: 3D assembly using polymer fluidic I/Os and solder electrical I/Os.	88
Figure 4.17	: Cross-sectional SEM image of 3D microfludic chip-to-chip bonding.	90
Figure 4.18	: (a.) Top-view IR-microscope image of 3D stack; (b.) Tilted IR-	
	microscope image of 3D stack.	90
Figure 4.19	: SEM images of 2-chip and 4-chip 3D stacks.	91
Figure 4.20	: 3D assembly and fluidic sealing.	91
Figure 4.21	: Experimental setup for fluidic testing of polymer pipe fluidic I/Os.	92
Figure 5.1:	Schematic of processor-on-processor 3D chip stack to be tested.	93
Figure 5.2:	Schematic of memory-on-processor 3D chip stack to be tested.	93
Figure 5.3:	Schematic of the top view of mask design with integrated (top layer).	94
Figure 5.4:	Schematic of wafer-level integration of microchannels, fluidic	
	through-silicon vias, electrical and fluidic I/Os, and integrated	
	platinum heaters/temperature sensors (top layer).	95
Figure 5.5:	Veeco Wycko microscope measurement of microchannel depth.	96
Figure 5.5:	Veeco Wycko microscope measurement of microchannel depth.	96
Figure 5.6:	Schematic of top view of mask design with integrated (bottom chip).	97
Figure 5.7:	Photograph of the bottom layer in the 3D stack.	97
Figure 5.8:	Schematic of wafer-level integration of microchannels, fluidic	
	TSVs, fluidic inlets/outlets, platinum heaters, and copper bonding	
	pads (bottom layer).	98

xiv

Figure 5.9:	SEM image of microchannels in the 3D chip stack bottom layer,	
	where inlet and outlet channels are on the left and right sides of the	
	microchannel heat sink (middle).	99
Figure 5.10:	Photograph of the multi-layer 3D stack.	100
Figure 5.11:	Schematic representation of the multi-layer 3D stack.	100
Figure 5.12:	Photograph of nanoports and fluidic tubing attached to the 3D	
	stack for fluidic continuity testing.	101
Figure 5.13:	SEM images of (left) microchannel heat sink and (right) integrated	
	thin-film platinum resistors and electrical I/Os.	102
Figure 5.14:	SEM images of (left) microchannel heat sink and (right) integrated	
	thin-film platinum resistors and electrical I/Os.	102
Figure 5.15:	(a.) Change of DC resistance of the heater as a linear function of	
	Temperature (ACHS).	103
Figure 5.15:	(b.) Change of DC resistance of the heater as a linear function of	
	Temperature (MCHS).	104
Figure 5.16:	Experiential setup for measuring multi-layer electrical connectivity.	105
Figure 5.17:	Schematic of the thermal test-bed experiential setup.	106
Figure 5.18:	Photograph of the thermal test-bed experiential setup.	107
Figure 5.19:	Schematic of ACHS and MCHS thermal testing setup.	108
Figure 5.20:	Heat sink volume comparison between the Intel i7 air-cooling	
	heat sink and the microchannel heat sink.	109
Figure 5.21:	On-chip temperature rise as a function of localized heating power	
	under various DI water flow rates (heating area of ~ 0.36 cm ²).	111
Figure 5.22:	Pressure drop as a function of flow rate for a single microfluidic	
	chip.	113

Figure 5.23	Pressure drop as a function of flow rate for a 3D stack of	
	microfluidic chips (two layers).	113
Figure 5.24	Top view of a SEM image of an array of micro pin-fins.	115
Figure 5.25	Top view SEM image of a micro pin-fin heat sink etched onto the	
	back side of a silicon chip. Inlet and outlet channels that are used	
	for routing cooling fluid to the pin-fin heat sink are shown on the	
	left and right sides of the pin-fin heat sink.	116

SUMMARY

Heat removal technologies are among the most critical needs for threedimensional (3D) stacking of high-performance chips. This research reports a 3D integration platform that can support the heat removal requirements for 3D integrated circuits containing high-performance microprocessors.

This work shows the use of wafer-level batch fabrication to develop advanced electrical and fluidic three-dimensional interconnect networks in a 3D stack. Fabrication results are shown for the integration of microchannels and electrical through-silicon vias (TSVs). A compact physical model is developed to determine the design trade-offs for microchannel heat sink and electrical TSV integration. An experimental thermal test-bed for evaluating a 3D inter-layer liquid cooling platform is developed. Experimental thermal testing results for an air-cooled chip and a liquid-cooled chip are compared. Microchannel heat sink cooling shows a significant junction temperature and heat sink thermal resistance reduction compared to air-cooling. The on-chip integrated microchannel heat sink, which has a thermal resistance of 0.229 °C/W, enables cooling of >100W/cm² per tier, while maintaining an average junction temperature of less than 50°C. Cooling liquid is circulated through the 3D stack (two layers) at flow rates of up to 100 ml/min.

The ability to assemble chips with integrated electrical I/Os (density of $\sim 1600/\text{cm}^2$) and fluidic I/Os at each strata interface is demonstrated using various assembly and fluidic sealing techniques.

xvii

CHAPTER 1

INTRODUCTION AND BACKGROUND

The silicon integrated circuit (IC) has served as the foundation of high performance computing, been the main driver of the information revolution, revolutionized the electronics industry, and has had a monumental impact on our society. Following the invention of the transistor in 1947 [1.1] and the subsequent invention of the integrated circuit in 1958 (Figure 1.1a) [1.2-1.4], Gordon Moore predicted that the number of transistors that can be integrated onto a microchip would double approximately every two years (Figure 1.2) [1.5-1.6].



Figure 1.1: (a.) The first integrated circuit. (b.) Intel's Itanium Poulson microprocessor containing over 3.1 billion transistors.

From manufacturing a chip with hundreds of transistors in the 1960s to manufacturing chips with over a billion transistors in the 2000s (Figure 1.1b), the semiconductor industry has kept pace with Moore's Law for over 50 years.



Figure 1.2: Moore's Law Data [1.6].

1.1. Thermal Management Challenges in High-performance ICs

1.1.1. Increasing Power Density

While Moore's Law and device scaling has provided higher functionality and performance, increased device density has also historically resulted in increased power dissipation and increased operating temperature. Chip operating temperature is a major determinate of semiconductor device reliability, and data shows that more than 50% of integrated circuit failures are related to thermal issues [1.7, 1.8]. Figure 1.3 shows historical data of how increasing power densities have accompanied successive microprocessor technology generations [1.9]. With respect to implications for on-chip interconnects, as temperature increases, the resistivity of interconnects increases, causing

decreased bandwidth and higher resistive losses as well as lower reliability due to electromigration.



Figure 1.3: Historical power density of various commercial microprocessors [1.9].

1.1.2. Static Power

Elevated temperatures can cause circuit timing related issues, on-chip temperature gradients, and increased leakage power consumption. Figure 1.4 depicts how leakage power substantially increases as chip temperature increases. At elevated temperatures, leakage power can account for over 50% of the power dissipation in an integrated circuit [1.7]. Thus, there is clear motivation to operate at lower temperature for highly scaled CMOS circuits.



Figure 1.4: Chip power consumption as a function of temperature; leakage power dependence on chip temperature [1.7].

1.1.3. Increasing Heat Sink Size

Although transistors have continued to scale smaller, thermal interconnects (interconnects responsible for rejecting heat to the ambient) have scaled inversely with CMOS technology. In order to achieve smaller junction-to-ambient thermal resistance and to maintain constant junction temperature with increasing power, the mass and volume of conventional air-cooling heat sinks have progressively increased with each new microprocessor generation [1.10]. As thermal interconnects have scaled larger, thermal interconnects impose limits on system size, chip packing efficiency, and interconnect length between chips. Figure 1.5 outlines how the heat sink volume of various Intel microprocessors has increased over time. The figure also illustrates how, in order to maintain junction temperature with increasing processor power dissipation, heat

sink size and fan power have increased [1.10]. Figure 1.6 illustrates an example of the size difference in heat sinks for two Intel Pentium processors from different technology generations.



Figure 1.5: Graph of increasing heat sink volume and heat sink fan power for various Intel microprocessors [1.10].



Figure 1.6: Pentium Overdrive "chip and heat sink" from 1993 (bottom left); standard Pentium 4 "heat sink only" from 2005 (top right).

1.1.4. Heat Sink Thermal Resistance

Figure 1.7 shows a schematic of a conventional air-cooling heat sink for a flip chip package. According to the International Technology Roadmap for Semiconductors (ITRS), the projected junction-to-ambient (ja) thermal resistance of an adequate heat sink for cooling high-performance microprocessors at the 14nm technology node should be approximately 0.2°C/W [1.11]. The overall thermal resistance (°C/W), R_{th}, of the heat sink and silicon chip shown in Figure 1.7 can be estimated as follows:

$$R_{th-ja} \approx R_{Si} + R_{TIM} + R_{sp} + R_{heat\sin k}$$

where R_{Si} , R_{TIM} , R_{sp} , and $R_{heatsink}$ are the thermal resistances of the silicon chip, the thermal interface materials (TIMs), the heat spreader, and the heat sink [1.12].

Yet, even when using the best available materials for the various thermal interconnects between the silicon die and the ambient (the heat sink, heat spreader, and thermal interface materials (TIM)), the sum of the thermal resistances typically associated with these thermal interconnects has a lowest attainable thermal resistance of approximately 0.5° C/W [1.13].



Figure 1.7: Schematic of thermal resistances associated with a conventional air-cooling heat sink.

1.1.5. Microprocessor Clock Frequency

Approximately during the past decade, state of the art microprocessors approached the heat flux air cooling limit of ~100 W/cm²[1.14]. Because of this thermal management limit (as well as other constraints), historic clock frequency scaling slowed and has been relatively flat since 2004, as shown in Figure 1.8 [1.14].



Figure 1.8: Frequency vs. Year for various Intel microprocessors [1.14].

To address power dissipation challenges, microprocessor design has shifted from single-core to multi-core processors [1.15]. Furthermore, having multiple strategically

placed cores on a single chip helps to mitigate thermal issues [1.16], as heat can be more evenly spread across the chip surface. Moreover, processor performance has continued to improve at a faster rate than memory access time, limiting overall system performance improvement. As the number of cores increases, issues of memory access time, bandwidth density, and off-chip interconnect latency become more challenging [1.15, 1.17].

1.2. Three-dimensional Integrated Circuits

The continuation of exponential improvement in productivity and performance by transistor scaling is becoming challenging. According to researchers, physical limits of atomic structures, power density limits, and thermal management requirements could cause Moore's Law and the accustomed rates of device scaling to substantially decrease and eventually come to an end [1.18-1.25].

3D integration is a promising technology that will extend Moore's Law in the "zdirection" by vertically integrating multiple layers of active electronic circuits into a single circuit. 3D integrated circuits offer many advantages including increased device density, shorter interconnect distances, system performance enhancements, decreased system form factor, and integration of heterogeneous technologies in the same chip stack.

Advances in through-silicon via (TSV) technology, wafer thinning, fine-pitch interconnections, and bonding have enabled stacking of multiple chips to achieve system performance enhancements [1.26-1.32]. Figure 1.9 shows examples of 3D integrated circuits implemented using electrical TSVs and wire-bonding.



Figure 1.9: Examples of 3D integration using electrical TSVs (left) [1.27] and wire bonding (right) [1.26].

Furthermore, because 3D technology can enable the integration of memory layers onto the processor chip, slower off-chip electrical interconnects can be eliminated and replaced with high-bandwidth, low-latency vertical interconnections [1.17]. Consequently, processor-memory interconnections which once required tens of millimeters of wire can be connected vertically using electrical TSVs, which are only tens of microns in vertical length [1.17]. As a result, chip-to-chip interconnection performance and bandwidth are significantly improved.

1.3. Thermal Challenges in Three-dimensional Integrated Circuits

Although there are many advantages to 3D integration, one of the most significant challenges is heat removal. To date, only low-power commercial products have been able to exploit the advantages of the improved performance and increased device packing density realized by the three-dimensional (3D) stacking of chips. Just as increased power density causes thermal management challenges in 2D integrated circuits, heat removal is

even more challenging in 3D integrated circuits because 3D technology enables much denser device integration.

1.3.1. Challenges of Conventional Air-cooling Heat Sinks

While stacking low-power memory chips can be thermally managed by conventional cooling methods [1.33], 3D chip stacks which contain multiple high-performance processors have a power density which exceeds the heat removal capability of conventional cooling techniques [1.34]. Because high-performance chips are projected to dissipate more than 100W/cm² (Table 1.1, Figure 1.10) [1.11], when such chips are stacked, the challenges in power delivery and cooling become greatly exacerbated. For example, for a two-chip stack of high-performance processers that dissipate 100W/cm², the power dissipation doubles from that of a single high-performance chip and the heat flux increases to 200W/cm² (Figure 1.11), which exceeds the conventional air-cooling heat sink capability [1.14].

With the addition of each device layer, the total power density and maximum junction temperature in the 3D stack increase. According to the ITRS, the number of high-performance chips (for high-performance applications) in a 3D chip stack will increase to 7 chips per stack and 10 chips per package by the year 2022 [1.11], as shown in Table 1.2.

Table 1.1: ITRS projections for high-performance chip junction temperature and power dissipation [1.11].

ITRS Data (High Performance Chips)		2013	2015	2017	2019	2021
Single Chip Junction Temperature (°C)		80	80	75	70	70
Dissipated Power (W)		149	143	130	130	130



Figure 1.10: ITRS projected power dissipation for high performance and cost performance microprocessors [1.11].

Table 1.2: ITRS projections for the maximum number of high-performance dice in a 3D chip stack and in a 3D package [1.11].

ITRS Data (High Performance Chips)		2012	2013	2014	2015	2016
Max Number of Stacked Die (TSV)		-	-	-	2	2
Number of Dice in Package		7	8	8	8	9
	2017	2018	2019	2020	2021	2022
Max Number of Stacked Die (TSV)	3	4	5	6	7	7
Number of Dice in Package		9	10	10	10	10



Figure 1.11: A schematic illustrating how heat flux increases from 2D to 3D circuits.

Furthermore, when using a conventional air-cooling approach, heat flux from chips in the 3D stack has to overcome a larger thermal resistance by traveling through a longer conductive path in order to be dissipated through the heat sink [1.35]. Moreover, because the heat generated in one stratum must travel through several other strata before dissipating into the heat sink, the temperature rise in one stratum influences temperature rise in other strata in the 3D stack [1.36].

If conventional air cooling techniques will not meet the power density, heat flux, and thermal resistance requirements for a single high-performance chip, as shown in Figure 1.7, conventional cooling will not be appropriate for a 3D stack of chips which has multiple high-performance processors (Figure 1.12) [1.34].

To further illustrate disadvantages of cooling a high-performance 3D chip stack with air-cooling, Figure 1.13 shows a 3D processor-memory stack where the processor chip is on top and the memory chip is on bottom. Stacking the processor chip, which dissipates more power, on the top layer is the best case thermally, as the processor is directly under the heat sink. However, using this approach makes it challenging to deliver the power required by a high-performance processor. Since the processor chip requires more I/O (input/output) connections than the memory chip, more electrical through-silicon vias (TSVs) will need to be routed through the memory chip consuming valuable memory chip area and making memory design and layout more difficult and less efficient.



Figure 1.12: Schematic of thermal resistances associated with a conventional air-cooling heat sink.



Figure 1.13: 3D thermal management using conventional air cooling.

In an alternative approach, the processor can be placed on the bottom of the 3D stack and the memory chip(s) on top. Because the memory chip has fewer I/Os, fewer TSVs are needed than in the previous case. Yet, placing the processor on the bottom of the stack is thermally disadvantageous, as the processor dissipates much more power than

the memory chip(s). With the addition of each memory chip, the thermal resistance is increased between the bottom device layer and the heat sink, limiting the number of chips that can be stacked.

1.3.2. Advanced Cooling of High-performance 3D Integrated Circuits

Thermal issues are increasingly becoming a device and system design concern, as researchers suggest that thermal management of high performance microprocessors is pushing the limits of air cooling, especially for applications such as servers used in data centers which contain high-performance processors that have high power dissipation and high density packaging [1.10, 1.14, 1.37]. Because the heat generated from multiple layers increases the potential of electrical failure [1.17, 1.38] and because the maximum allowable average power density is constrained by the limitation of heat removal capability of the heat sink [1.8], the increased device density as a result of 3D chip stacking requires innovative thermal management strategies. Thus, a significant amount of research is being focused on creating advanced cooling technologies.

Liquid cooling using microchannels has been identified as a promising cooling solution to meet the thermal management requirements of future high-performance microprocessors due to the superior thermal properties of liquid compared to air. Figure 1.14 shows a comparison of heat transfer coefficients of various thermal buses, which shows that the heat transfer coefficient of removing heat through microchannel cooling is a higher capacity heat removal solution compared to forced air cooling [1.12].



Figure 1.14: Heat transfer coefficient of various thermal buses [1.12].

Tuckerman and Pease were the first to demonstrate an on-chip, integrated microchannel heat sink which showed the ability to remove a heat flux of 790 W/cm² with a maximum chip temperature rise of 71°C [1.39]. Although a number of researchers have explored the advantages of using liquid cooling to solve thermal management challenges [1.12, 1.40-1.43], there are many unknowns for its implementation for 3D integrated systems. Some of the unknowns for liquid cooling using microchannels include fabrication of an on-chip microfluidic heat sink and integration of electrical through-silicon vias (TSVs), where to place fluidic I/O interconnects for 3D chips, how to supply fluid to and extract fluid from microchannels embedded in a 3D stack, how to seal fluidic I/O interconnections at the interface of each strata, and how to assemble 3D ICs with microfluidic functionality.

1.4. Summary of Research Objectives

The objective of this research is to investigate and report the configuration, fabrication, assembly, and experimental results of a 3D integration platform that can support the heat removal requirements for high-performance chips. A novel microchannel heat sink liquid cooling scheme for 3D integrated circuits (ICs) and the fabrication processes necessary for integrating microchannels and electrical TSVs will be outlined. The on-chip integrated microchannel heat sinks, which have a thermal resistance of 0.229 °C/W, enable cooling of >100W/cm² per tier at an average junction temperature of less than 50°C. Compact physical modeling is used to determine the design trade-offs for microchannel heat sink and electrical TSV integration.

Fluidic input/output (I/O) interconnect technologies that enable fluidic connectivity of the 3D microfluidic network are demonstrated. Three distinct fluidic I/O technologies and assembly methods used for 3D chip stacking and fluidic sealing are demonstrated. Simultaneous fabrication of electrical and fluidic I/Os is demonstrated, and an electrical I/O density of $\sim 1600/\text{cm}^2$ is achieved.

An experimental thermal measurement test-bed for evaluating a 3D inter-layer liquid cooling platform is developed. The experimental thermal measurements demonstrate the cooling of chips which dissipate a heat flux of 100W/cm²; electrical and fluidic interconnection between layers is also demonstrated. Cooling liquid is circulated through the 3D stack at flow rates of up to 100 ml/min.

1.5. Dissertation Outline

The organization of the research in this dissertation is described as follows:
Chapter 2 describes the wafer-level batch fabrication and micromachining technologies used to fabricate the necessary electrical and microfluidic interconnects for the proposed 3D inter-layer cooling platform. Each silicon die of the 3D stack contains the following features: 1) a monolithically integrated microchannel heat sink, 2) through-silicon fluidic vias (TSFV) used for fluidic routing in the 3D stack, and 3) solder bumps (electrical I/Os) and microscale controlled collapse chip connection (C4) pipes (fluidic I/Os) on the side of the chip opposite to the microchannel heat sink.

Additionally, compact physical modeling is used to analyze the impact of microchannel geometry and fluid flow rates on thermal resistance and pressure drop of the 3D systems. Additionally, compact physical modeling is also used to explore the electrical TSV performance and microchannel heat sink cooling trade-offs when integrating microchannel heat sinks and electrical TSVs in a 3D chip stack.

- Chapter 3 describes fabrication and process integration techniques for three distinct fluidic I/O technologies including a C4 pipe fluidic I/O, an air-gap C4 fluidic I/O, and a polymer pipe fluidic I/O interconnect technology. Fluidic testing is performed to verify the reliability of the fluidic I/O interconnect structures. The advantages and disadvantages of the three fluidic I/O technologies are discussed.
- Chapter 4 discusses the flip-chip die-to-substrate and die-to-die bonding processes that enable the integration of electrical and fluidic components in a 3D chip stack.

- Chapter 5 outlines the thermal and fluidic analysis and testing results of the onchip microhannel heat sink. Experimental results for an air-cooled chip and a liquid-cooled chip are compared. Preliminary fabrication results for an alternative pin-fin heat sink technology are demonstrated, and the advantages of the pin-fin heat sink compared to the microchannel heat sink technology are discussed.
- Chapter 6 presents the conclusion and opportunities for future work.

CHAPTER 2

DEVELOPMENT OF AN INTER-LAYER LIQUID COOLING PLATFORM FOR THERMAL MANAGEMENT OF 3D INTEGRATED CIRCUITS

2.1. Development of an Inter-layer Liquid Cooling Platform for Thermal Management of 3D Integrated Circuits

Cooling multiple stacked high-power chips presents significant thermal challenges. Thus, revolutionary advanced cooling technologies are necessary to provide adequate cooling and enable circuit designers to have full flexibility when designing processor-memory and processor-processor 3D chip stacks. Using integrated microchannel heat sinks to cool high-performance 3D chip stacks is discussed in [2.1], where experimental results show that single chips cooled by microchannel heat sinks exhibit a junction-to-ambient thermal resistance of 0.24°C/W. Thus, using inter-layer microchannel heat sinks to cool chips at any tier in the 3D stack allows higher chip stacks, enabling more functionality in a given system foot print.

Figure 2.1a outlines a 3D stack configuration where the processor is on the bottom of the 3D stack. Although this approach would be thermally challenging using air cooling, the integrated microchannel heat sink on the back side of the processor enables sufficient cooling of the processor layer, while enabling most efficient power delivery to the processor. When stacking multiple high performance processors (Figure 2.1b), each die in the stack can be thermally isolated. One of the most significant aspects of this

technology is that multiple processor and memory chips can be stacked without having to place the high power chips on the top of the stack, which would be the best thermal solution when using conventional air cooling but not the most effective configuration for power delivery to the processor chip [2.2].



Figure 2.1: 3D IC (a) processor-memory and (b) processor-processor thermal management approaches using a microchannel heat sink inter-layer liquid cooling.

The microchannel liquid cooling scheme shown in Figure 2.2 can be used for cooling 3D chip stacks that contain high-performance processors. The cooling scheme reduces the overall thermal resistance of the cooling system, removes thermal resistances associated with TIMs, reduces chip cooling hardware size from inches to microns, and enables cooling of >100W/cm² of each high-power density chip [2.3-2.4].

2.2. 3D Inter-layer Liquid Cooling Platform Process Development and Integration of Microchannel Heat Sink, Electrical TSVs, Fluidic TSVs, Electrical I/Os, and Fluidic I/Os

Figure 2.3 outlines the processing steps necessary for fabrication of the features that enable the electrical and fluidic networks shown in Figure 2.2. Following back-endof-the-line (BEOL) processing, microchannels and fluidic TSVs are formed into the back side of the wafer by inductively coupled plasma (ICP) etching (Figure 2.3a, 2.3b). The microchannels can be capped using the sacrificial polymer process described in [2.1] or by the silicon-to-silicon bonding process described in [2.5] (Figure 2.3c). Next, TSVs are etched in the wafer using ICP etching, and copper is electroplated in the TSVs as shown in [2.6] (Figure 2.3d, 2.3e). Figure 2.4 shows 200µm deep, 100µm wide polymer-capped microchannels with integrated 50µm diameter copper electrical TSVs. Figure 2.5 shows 170µm tall, 100µm wide silicon-capped microchannels with integrated 50µm diameter copper electrical TSVs. This work is the first 3D cooling research to demonstrate the integration of microchannel heat sinks and electrical TSVs (Figures 2.4-2.5), as shown in [2.3, 2.7].

Subsequently, polymer sockets, which enable assembly of the electrical and fluidic I/Os are fabricated by spinning on and patterning a photo-definable polymer as described in [2.3] (Figure 2.3f). Electrical I/O interconnects (solder bumps) are fabricated by electroplating, and fluidic I/O interconnects (polymer or solder pipes) are fabricated patterning a photo-definable polymer [2.8] or electroplating solder [2.9]. Figure 2.6 shows an SEM image of integrated electrical and fluidic I/O interconnects

[2.9]. Fabrication processes for electrical and fluidic I/Os are described in more detail in Chapter 3.



Figure 2.2: Schematic of chip-scale microchannel heat sink cooling scheme for 3D integrated circuits.

Microscale fluidic interconnection between strata is enabled by through-wafer fluidic vias and fluidic I/O interconnects. Power delivery and signaling can be supported by the electrical interconnects (solder bumps and copper TSVs), and heat removal for each stratum can be supported by the fluidic I/Os and microchannel heat sinks. Chips can be aligned, stacked, and assembled; assembly processes are discussed in the Chapter 4.



Figure 2.3: Schematic of wafer-level integration of microchannels, fluidic through-silicon vias, silicon dioxide insulated electrical through-silicon vias, and electrical and fluidic I/Os to enable liquid cooling of high-performance 3D integrated systems.



Figure 2.4: Cross-sectional optical image and schematic of a polymer-capped microchannel heat sink and integrated electrical through-silicon vias [2.3, 2.7].



Figure 2.5: (a) X-ray image and (b) cross-sectional optical image of a silicon-capped microchannel heat sink and integrated electrical through-silicon vias [2.5].



Figure 2.6: SEM image of integrated electrical and fluidic I/O interconnects [2.9].

A thin-film platinum resistor is fabricated on the bottom side of each chip of the 3D stack (Figure 2.7). When applying a current source to the resistors, the resistors serve as a heating source to simulate heat dissipated by transistors and interconnects on a microprocessor. The change in resistance of the resistor is measured and used to calculate the change in chip temperature.



Figure 2.7: SEM images of (left) microchannel heat sink and (right) integrated thin-film platinum resistors and electrical I/Os.

2.3. Methods for Microchannel Heat Sink Capping

As shown in Figure 2.3, trenches can be etched into the back side of a silicon chip by inductively coupled plasma (ICP) etching (Figure 2.3a, 2.3b), thereby creating microchannels on the back side of the chip that can be used as an on-chip microchannel heat sink. Figure 2.8 shows SEM images of microchannels that have been etched onto the back side of a silicon chip. After trench etching, the microchannels must be capped to enable liquid to be circulated on the back side of the chip. Although various methods for microchannel heat sink fabrication have been explored, the three methods of microchannel capping investigated during this research are using a polymer overcoat [2.1], using silicon-to-silicon bonding to create a silicon cap [2.5, 2.9], and using adhesive bonding [2.5]. The silicon-to-silicon bonding and polymer overcoat methods will be discussed in the following sections.



Figure 2.8: Top view of microchannel heat sink and fluidic TSVs after silicon etching.

2.3.1. Silicon-to-Silicon Bonding

Microchannel capping can be achieved by direct, silicon-to-silicon (Si-Si) wafer bonding. Semiconductor wafer bonding can be classified by a process where two polished wafers can be adhered or bonded together at room temperature under atmospheric conditions without the use of an adhesive layer or other outside force. Bonding is established by molecular bonds between molecules on the surfaces of the two substrates. The adhesion forces responsible for enabling this process include Van der Waals forces, electrostatic coulomb forces, capillary forces, or hydrogen bridge bonds [2.10]. Initial bonding is performed at room temperature, where a weak adhesion is established between the substrates and subsequently followed by a high temperature annealing step where the strength of the weak bonds is increased by a factor of 10 and converted to strong permanent bonds by transforming the hydrogen bridge bonds across the interface via the reactions (shown in Figure 2.9) [2.11].



Figure 2.9: Chemical reactions during the hydrophilic bonding process.

Because direct bonding can require annealing temperatures >1000 °C to establish a permanent bond, for low-temperature processes, establishing appropriate surface conditions on the two substrates is necessary to achieve sufficient interface bonding forces during low-temperature annealing.

Activation of the two bonding surfaces by increasing the density of the silanol groups on the surface via plasma treatment helps to increase bonding strength, allowing strong hydrophilic bonding at annealing temperatures <400 °C [2.12-2.13].

For hydrophilic surfaces, the interface energy is equivalent to the number of silanol groups (Si-OH) at the initial surface. Therefore, activation of the surfaces by increasing the density of silanol groups would also increase the interface energy. This activation can be achieved using plasma treatment. It has been reported that strong hydrophilic bonding with Si can be achieved at a low annealing temperature <400°C by exposing wafers to a low pressure plasma prior to the bonding [2.12-2.13].

To cap the microchannels, a low temperature plasma-activated Si-Si direct bonding approach was implemented for capping microchannels. Figure 2.10 outlines the fabrication processing steps necessary to achieve wafer-to-wafer bonding. First, the surface of a wafer containing microfluidic channels and a 400µm double-side polished silicon capping wafer are dipped in RCA-1 solution (H₂O, NH₄OH, H₂O₂, mixed with a ratio of 5:1:1 respectively). RCA-1 solution removes organic impurities that may be present on the surface of the wafers and changes the surface on the silicon wafers to hydrophilic surfaces.

Next, the wafer surface is treated with O₂ plasma in a reactive ion etching (RIE) system and bonded at room temperature and in atmosphere. During the plasma activation process, oxidation occurs on the surface, which helps to facilitate high bonding strength [SB4]. After plasma treatment, the wafers are rinsed with DI water, and bonding can be performed by aligning and bonding the two wafers by hand at room temperature. Additionally, for better control of applied force and force distribution across the wafers during bonding, a wafer-to-wafer bonding tool may be used.



Figure 2.10: Schematic of process flow for capping microchannels via silicon-to-silicon bonding.

Finally, thermal annealing is performed at 400 °C for 12 hours, transforming the weak chemical bonds to strong chemical bonds. Silicon-capped microchannels were tested by examining the bonded wafer for voids and cracks using an infrared microscope, by destructive tests such as dicing, and by passing fluid through the microchannels at flow rates up to 100ml/min. Figure 2.11 shows a cross-sectional SEM image of a bonded silicon wafer containing microchannels and a silicon capping wafer. Wafers were consistently bonded with a yield of 70-80% using the Si-Si bonding technique [2.5].



Figure 2.11: Schematic of process flow for capping microchannels via silicon-to-silicon bonding.

2.3.2. Polymer Overcoat

A second method for capping microchannels is by using the sacrificial polymer process described in [2.1, 2.7]. After etching microchannel trenches into the back side of the wafer (Figure 2.12a), Unity sacrificial polymer (Promerus, LLC) is spin-coated on the wafer, filling the microchannels. Afterwards, mechanical polishing is performed to planarize the surface (Figure 2.12b). Next, 15 μ m of Avatrel 2090P polymer (Promerus, LLC) is spin-coated onto the wafer (Figure 2.12c). Finally, the Avatrel polymer is cured, and the Unity sacrificial polymer is thermally decomposed simultaneously in a nitrogenpurged furnace at low temperatures ($\leq 200^{\circ}$ C) (Figure 2.12d) [2.8]. Figure 2.4 shows a cross-sectional optical image of a sample after the previously described processing steps are completed. The non-optimized microchannels are 200µm tall and 100µm wide (Figure 2.4), and the copper TSVs have a 50µm diameter.



Figure 2.12. Schematic of capping microchannels using a polymer overcoat. (a)Microchannel trenches are etched into the back side of the wafer. (b)Spin coat and polish Unity sacrificial polymer. (c) Spin coat and pattern Avatrel polymer sockets as an overcoat layer. (d) Simultaneous curing of Avatrel polymer and thermal decomposition of sacrificial polymer.

2.4. Microchannel Heat Sink Design

2.4.1. Microchannel Heak Sink Theory

In [1.39], Tuckerman and Pease derived and experimentally verified the equations necessary to calculate the thermal resistance of a microchannel heat sink. A schematic of the microchannel heat sink concept demonstrated by Tuckerman and Pease is shown in Figure 2.13.



Figure 2.13: Schematic illustration of a microchannel heat sink concept demonstrated by Tuckerman and Pease [1.39].

The thermal resistance of a microchannel heat sink can be calculated using the sum of three resistances [1.39, 2.8]:

$$R_{th} = R_{cond} + R_{conv} + R_{heat},$$

where R_{cond} , the resistance due to conduction, is dependent on thermal conductivity of silicon, the distance between the bottom surface of a channel and the integrated circuits at the bottom side of the wafer, and the chip area. R_{conv} , the convective resistance, is dependent on the heat transfer coefficient of the cooling fluid and the area of the surfaces of the microchannels. The surface area of the channels is determined by the channel width, channel length, channel height, and the number of channels. R_{heat} , the resistance

due to heating of the cooling fluid, is determined by the temperature rise of the cooling fluid between the liquid inlet and outlet and the heat flux generated by the integrated circuit on the bottom side of the chip [1.39, 2.8, 2.14].

The conductive thermal resistance, R_{cond} , can be calculated by using the following equation:

$$R_{cond} = \frac{t}{k_{Si} A_{chip}},$$

where k_{Si} is determined by the thermal conductivity of silicon, A_{chip} is established by the area of the chip, and *t* is determined by the distance between the bottom surface of the microchannel and the integrated circuits at the front side on the chip.

The convective resistance, R_{conv} , is determined by the following equation:

$$R_{conv} = \frac{1}{hA_{channels}} ,$$

where the area of the microchannels is calculated as:

$$A_{channels} = n_c \cdot L_c \cdot (2H_c + W_c).$$

The surface area of the channels is determined by the channel width (W), channel length (L), channel height (H), and the number of channels (n). Consequently, the convective resistance is proportional to the channel height, channel width, and channel length.

For modeling and experimental purposes, deionized water is used as the cooling liquid. Assuming that DI water is at a laminar flow [2.15-2.16], the convective heat transfer coefficient h can be calculated by the following equation:

$$h = \frac{k_f N u_{fd}}{D_h}.$$

where k_f is the fluid thermal conductivity, and D_h is the hydraulic diameter of a microchannel:

$$D_h = \frac{2W_c H_c}{H_c + W_c}.$$

 Nu_{fd} is the Nusselt number of the fluid, a dimensionless heat transfer coefficient, which can be calculated by using the following equation:

$$Nu_{fd} = 8.235 (1 - 1.883\alpha + 3.767\alpha^2 - 5.814\alpha^3 + 5.361\alpha^4 - 2\alpha^5),$$

where α is the ratio of channel width to the channel height (W_c/H_c) [2.8, 2.14].

The third component, R_{heat} , can be calculated by using the following equation:

$$R_{heat} = \frac{\Delta T_{inlet/outle}}{Q} = \frac{1}{V \cdot \rho \cdot C_{p}},$$

where V is the liquid flow rate, ρ is the density of the cooling liquid, and C_p is the specific heat capacity of the cooling liquid.

According to the ITRS, the required heat sink thermal resistance for high performance microprocessors will be 0.2° C/W by the year 2018 [1.40], and the necessary chip junction temperature will continue to decrease (Table 2.1) [1.11].

ITRS Data (High Performance Chips)	2011	2012	2013	2014	2015	2016
Single Chip Junction Temperature (°C)	85	85	80	80	80	75
Dissipated Power (W)	161	158	149	152	143	130
	2017	2018	2019	2020	2021	2022
Single Chip Junction Temperature (°C)	75	75	70	70	70	70
Dissipated Power (W)	130	130	130	130	130	130

Table 2.1: ITRS data for high-performance microprocessors.

Figure 2.14 shows a summary of the equations necessary for microchannel heat sink analysis [1.39, 2.8, 2.14-2.16].

Equations	Symbol Legend
$R_{th} = R_{cond} + R_{conv} + R_{heat}$	R_{th} = Total thermal
$R_{cond} = \frac{t}{k_{Si} A_{chip}}$	$A_{channels} = area of$ channel
$R_{conv} = \frac{1}{hA_{channels}}$	h = heat transfer coefficient
$R_{i,ij} = \frac{\Delta T_{inlet/outle}}{1} = \frac{1}{1}$	$H_c = channel height$
$Q \qquad V \cdot \rho \cdot C_p$	L_c = channel length
$A_{channels} = n_c \cdot L_c \cdot (2H_c + W_c)$	$D_h =$ channel hydraulic diameter
$h = \frac{k_f N u_{fd}}{D}$	$Nu_{fd} = Nusselt number$
$D_h = \frac{2W_c H_c}{W_c H_c}$	ΔP_{ch} =pressure drop in microchannels
$H_c + W_c$	f = friction coefficient
$Nu_{fd} = 8.235 (1 - 1.883\alpha + 3.767\alpha^2 - 5.814\alpha^3 + 5.361\alpha^4 - 2\alpha^5)$	Re = Reynolds number
$1 \qquad (I + W / H)^2$	V = flow rate
$\Delta P_{ch} = \frac{1}{2} f \cdot \operatorname{Re} \cdot L_c \cdot \frac{\mu \cdot v \left(1 + w_c / H_c\right)}{n \cdot H_c \cdot W_c^3}$	C_p = specific heat
$f \mathrm{Re} = 64$	ρ =fluid density
$v = \frac{flow \ rate}{V} = \frac{V}{V}$	
overall cross – sectional area $n_{via}\pi(\frac{D_{via}^2}{4})$	$k_{si} = Si$ thermal conductivity
	$k_f =$ fluid thermal conductivity
	$n_c =$ number of channels
	Q = Power

Figure 2.14: Hydro-dynamic equations for microchannel heat sink theory.

2.4.2. Microchannel Heak Sink Geometry and Thermal Resistance

The channel geometry of a microchannel heat sink is a major component of determining the thermal resistance of a microchannel heat sink. Figure 2.15 shows a schematic of a microchannel heat sink.



Figure 2.15: Schematic illustration of a microchannel heat sink [1.39].

Matlab was used to calculate the thermal resistance of various heat sink configurations in order to determine the appropriate microchannel geometry for the proposed microchannel heat sink. The chip size used for modeling purposed was chosen to be 1cm x 1cm in area. To analyze the impact of the channel height on the heat sink thermal resistance, a fixed number of channels, a fixed fluid flow rate, and a fixed channel fin width were chosen. Thermal resistance values were generated for varying

microchannel heights of 100μ m to 400μ m, as shown in Figure 2.16, Table 2.2, Table 2.3, and Table 2.4.

Simulation results show that microchannel heat sink thermal resistance decreases as the channel height increases. As the channel height increases, the surface area that is in contact with the cooling fluid increase, causing the thermal resistance to decrease. Figure 2.16 also illustrates that the thermal resistance decreases as the channel fin width (W_w) decreases. This reduction in thermal resistance occurs because as the channel fin width is decreased, the number of microchannels that can be integrated into a given area increases, which enables more microchannel cooling capability.



Figure 2.16: Simulation results for thermal resistance as a function of channel height for various fixed channel fin widths.

Channel Height (µm)	Thermal Resistance (C/W)	Flow Rate (ml/min)	Number of Channels	Channel Width (µm)
100	0.628	100	51	100
150	0.466	100	51	100
250	0.322	100	51	100
350	0.259	100	51	100
400	0.239	100	51	100

Table 2.2: Summary of simulation results for thermal resistance as a function of channel height (channel width = 100μ m).

Table 2.3: Summary of simulation results for thermal resistance as a function of channel height (channel width = 50μ m).

Channel Height (µm)	Thermal Resistance (C/W)	Flow Rate (ml/min)	Number of Channels	Channel Width (µm)
100	0.292	100	101	50
150	0.239	100	101	50
250	0.197	100	101	50
350	0.176	100	101	50
400	0.169	100	101	50

Table 2.4: Summary of simulation results for thermal resistance as a function of channel height (channel width = 150μ m).

Channel Height (µm)	Thermal Resistance (C/W)	Flow Rate (ml/min)	Number of Channels	Channel Width (μm)
150	0.772	100	34	150
250	0.512	100	34	150
350	0.388	100	34	150
400	0.349	100	34	150

2.4.3. Microchannel Heak Sink Geometry and Pressure Drop

The channel geometry of a microchannel heat sink is also a determinant of the pressure drop in the heat sink. The pressure drop in the system can be calculated as a

function of the chip power (heat flux) to be removed based on the hydrodynamic equations found in Figure 2.14,

$$\Delta P_{ch} = \frac{1}{2} f \cdot \operatorname{Re} L_c \cdot \frac{\mu \cdot V (1 + W_c / H_c)^2}{n \cdot H_c \cdot W_c^3}.$$

For modeling purposes, it is assumed that DI water is used as the cooling fluid and a fully developed laminar flow is present in the microchannels, and a fixed chip area of 1cm x 1cm is chosen.

To analyze the impact of the channel height on the heat sink pressure drop, a fixed number of channels, a fixed fluid flow rate, and a fixed channel fin width were chosen. Pressure drop values were generated for varying microchannel heights of 100µm to 400µm, as shown in Figure 2.17 and Table 2.5.



Figure 2.17: Simulation results for thermal resistance as a function of channel height (channel width = $100\mu m$).

Channel Height (µm)	Pressure Drop (kPa)	Flow Rate (ml/min)	Number of Channels	Channel Width (µm)
100	82.8	100	49	100
150	39.6	100	49	100
200	25.5	100	49	100
300	14.7	100	49	100
400	10.4	100	49	100

Table 2.5: Summary of simulation results for pressure drop as a function of channel height (channel width = 100μ m).

To analyze the impact of channel width on the heat sink pressure drop, a fixed channel height and a fixed fluid flow rate were chosen. Pressure drop values were generated for varying microchannel widths of 50µm to 150µm, as shown in Figure 2.18, and Table 2.6.



Figure 2.18: Simulation results for pressure drop as a function of channel width (channel height = $350 \mu m$).

Channel Height (µm)	Pressure Drop (kPa)	Flow Rate (ml/min)	Number of Channels	Channel Width (µm)
350	43.4	100	99	50
350	23.3	100	62	80
350	12.2	100	49	100
350	7.9	100	37	130
350	6.2	100	32	150

Table 2.6: Summary of simulation results for pressure drop as a function of channel width (channel height = 350μ m).

Results of the Matlab simulation show that pressure drop can be decreased by increasing the channel width or by increasing the channel height, as shown in Table 2.5 and Table 2.6. However, it is important to note that if the channel width is increased, the number of channels that can be fabricated in a given area decreases. Consequently, the overall thermal resistance of the microchannel heat sink increases.

It is also important to note that, when increasing the channel height while having a fixed channel width, as pressure drop decreases, thermal resistance decreases, as shown in Figure 2.17 and Table 2.5.

Figure 2.19 shows a graph of microchannel heat sink thermal resistance as a function pressure drop. For this simulation, the channel width is fixed, and the microchannel height varies from 100 μ m to 400 μ m. The graph shows that as pressure drop decreases, the microchannel heat sink thermal resistance decreases. This result is expected, as channel height (H_c) is inversely proportional to ΔP_{ch} , R_{cond} and R_{conv} .



Figure 2.19: Simulation results for thermal resistance as a function of pressure drop.

Channel Height (µm)	Pressure Drop (kPa)	Flow Rate (ml/min)	Number of Channels	Channel Width (µm)	Thermal Resistance (C/W)
100	82.8	100	49	100	0.456
150	39.6	100	49	100	0.393
200	25.5	100	49	100	0.347
300	14.7	100	49	100	0.291
400	10.4	100	49	100	0.263

Table 2.7: Summary of simulation results for thermal resistance as a function of pressure drop.

Using the equations in Figure 2.14 to determine the thermal resistance and pressure drop for various microchannel heat sink configurations and by using the ITRS projected heat sink thermal resistance required for high-performance chips, the dimensions of proposed microchannel heat sink design were calculated, as shown in Table 2.8. Each chip in the 3D stack has 39, 80µm wide, 350µm tall microchannels.

SEM images of the fabricated microchannel heat sink are shown in Figure 2.8, and a schematic of the mask layout of the microchannels is shown in Figure 2.20. A more detailed discussion of the heat sink thermal resistance and pressure drop values that were measured during thermal and fluidic testing will be discussed in more detail in Chapter 5.

Table 2.8: Microchannel heat sink dimensions for proposed 3D cooling scheme (fluid flow rate = 100 ml/min).

Microchannel Heat Sink Dimensions			
Channel Width	80 um		
Channel Height	350 um		
Fin Width	80 um		
Thermal Resistance	0.229 °C/W		



Figure 2.20: Schematic illustration of proposed microchannel heat sink design.

2.5. Microchannel Heat Sink and Electrical TSV Co-design

In the proposed 3D inter-layer liquid cooling technology, while microchannels are needed for cooling, electrical TSVs are needed for signaling and power delivery in the

3D system. For 3D technology, usually wafers are thinned down in order to make electrical TSV fabrication easier. However, as previously discussed, microchannel heat sink cooling capability is a function of channel geometry. Cooling capability increases with increasing channel surface area in contact with the cooling liquid. Consequently, there are tradeoffs when co-designing electrical TSVs and microchannel heat sinks for integration in 3D chip stacks.

2.5.1. Microchannel Heak Sink Geometry and Electrical TSV Density

According to the ITRS, the minimum TSV pitch will continue to decrease, the minimum die thickness will continue to decrease, and the number of electrical TSVs in a high-performance processor will continue to increase through the year 2022 [1.11], as shown in Table 2.9. Consequently, when using a microchannel heat sink as the chip cooling solution, it is important to use a channel geometry that allows a sufficient number of electrical TSVs to be fabricated in the chip.

ITRS Data (High Performance Chips)	2011	2012	2013	2014	2015	2016
TSVs (x1000)	-	-	-	-	5	5
Minimum TSV Pitch (µm) (die to die)	4	3.8	3.6	3.4	3.3	3.1
TSV Maximum Aspect Ratio	10	10	10	10	10	10
Minimum Die Thickness (µm) (TSV)	-	-	-	-	25	20
Minimum Die Thickness (µm) (Side by Side)	100	100	100	100	100	100
	2017	2018	2019	2020	2021	2022
TSVs (x1000)	6	6	7	7	8	8
Minimum TSV Pitch (µm) (die to die)	2.9	2.8	2.7	2.5	2.4	2.3
TSV Maximum Aspect Ratio	10	10	10	10	10	10
Minimum Die Thickness (µm) (TSV)	20	18	18	15	15	10
Minimum Die Thickness (µm) (Side by Side)	100	100	100	100	100	100

Table 2.9: ITRS projections for high-performance chips: number of electrical TSVs, minimum TSV pitch, maximum TSV aspect ratio, and minimum die thickness [1.11].



Figure 2.21: ITRS projections for number of electrical TSVs per high-performance chip [1.11].

As the channel fin width increases, more electrical TSVs can be fabricated in the channel fin (also known as the microchannel wall, W_w), as shown in Figure 2.2. However, as the width of the microchannel fins increase, the number of microchannels that can be fabricated in a given area decreases, which results in less cooling capability.

Based on channel geometrical considerations, Matlab code was used to calculate the number of electrical TSVs that can be fabricated in silicon channel walls of a microchannel heat sink, as a function of electrical TSV diameter. The height of the microchannel channels and thickness of the wafer range from 200 μ m to 400um. The chip width (*W*) is assumed to be 1cm, and the chip length (*L*) is assumed to be 1cm.



Figure 2.22: Schematic illustration of a wafer with integrated electrical TSVs and a microchannel heat sink [2.17].

The width of the microchannel wall (W_W) can be expressed as a function of TSV parameters as,

$$W_W = 50 + P_{TSV} \cdot (N_{TSVColumns} - 1) + D_{TSV},$$

where P_{TSV} is the TSV pitch, D_{TSV} is the TSV diameter, and $N_{TSVcolumns}$ represents the number of TSV columns in the microchannel wall. ($N_{TSVcolumns} \ge 1$) [2.17]. The first term in the above equation, 50µm, is an assumed constant value that is used to establish the clearance between the edge of the outermost column of TSVs and the edge of the microchannel wall, as shown in see Figure 2.23. P_{TSV} is assumed to be 1.5(D_{TSV}).

The total number of TSVs, N_{TSVs} , that can be fabricated on a 1 cm^2 chip in the microchannel heat sink fins can be estimated as

$$N_{TSVs} = N_{TSVColumns} \cdot (n+1) \frac{L}{P_{TSV}},$$

where n is the total number of microchannels. , which can be calculated using the expression

$$n = \frac{W - W_W}{W_W + W_C}$$



Figure 2.23: Schematic of microchannel fin with integrated electrical TSVs.

Figures 2.24 and 2.25 show simulation results for microchannel heat sink thermal resistance and pressure drop as a function of TSV diameter (20µm diameter and 5µm diameter), number of TSVs columns per microchannel fin, and varying microchannel heights (wafer thickness) that vary from 200µm to 400µm.

The simulation results are summarized in Table 2.10. The simulation results show that as the width of microchannel fin increases, the number of columns of electrical TSVs can also be increased, which causes the microchannel heat sink thermal resistance and pressure drop to increase. This increase occurs because, as the number of TSV columns increases, the width of the channel fin increases. For a fixed microchannel width, this results in a decreased number of microchannels available for cooling. A decreased number of microchannels leads to a decrease in the surface area in contact with the cooling fluid. Furthermore, a decrease in the number of microchannels also increases

the pressure drop, as the number of channels in inversely proportional to the pressure drop in the microchannel heat sink.

Additionally, an increase in the diameter of the electrical TSVs also causes an increase in microchannel fin width. Consequently, the thermal resistance and pressure drop increase, as the number of microchannels available for cooling decreases. An increase in thermal resistance and pressure drop are also observed when the microchannel height and wafer thickness increase due to the impact of channel geometry on thermal resistance and pressure drop, as previously discussed.



Figure 2.24: Electrical TSV density as a function of channel fin width and electrical TSV diameter (20um).



Figure 2.25: Electrical TSV density as a function of channel fin width and electrical TSV diameter (5um).

The results from the analysis shown in Table 2.10 illustrate that the electrical TSV density required for future high-performance chips, based on ITRS projections, can be achieved when using a microchannel heat sink cooling solution. Consequently, the limiting factors for integrating microchannel heat sinks and electrical TSVs will be electrical TSV aspect ratio and the amount of chip area that will be dedicated to electrical TSVs.

Number of TSVs	TSV Diameter (μm)	Number of TSV Columns	Number of Channels	Channel Fin Width (µm)	Channel Width (µm)
360,000	5	5	53	85	100
600,000	5	10	44	122.5	100
760,000	5	15	37	160	100
880,000	5	20	32	197.5	100
150,000	10	5	44	120	100
226,670	10	10	33	195	100
270,000	10	15	26	270	100
293,330	10	20	21	345	100
86,667	15	5	38	155	100
120,000	15	10	26	267.5	100
140,000	15	15	20	380	100
151,110	15	20	16	492.5	100
56,667	20	5	33	190	100
73,333	20	10	21	340	100
85,000	20	15	16	490	100
86,667	20	20	12	640	100

Table 2.10: Electrical TSV density as a function of channel fin width and electrical TSV diameter.

Regarding aspect ratio, the minimum aspect ratio of 10:1 is projected for electrical TSVs integrated in high-performance chips, as shown in Table 2.9 [1.11]. In the literature, a copper filled TSV with an aspect ratio of 49:1 has been demonstrated in a thinned silicon wafer [2.17]. The analysis that was done in this research shows electrical TSV density results for electrical TSVs with diameters of 5µm, 10µm, 15µm, and 20µm and for microchannel channel heights and wafer thickness ranging from 200µm to 400µm. For the smallest diameter electrical TSV used in this analysis, 5µm diameter TSVs, the aspect ratio would be 40:1 when using a 200µm thick silicon wafer and 80:1 when using a 400µm thick wafer. Consequently, for cooling solutions that require thicker chips, electrical TSVs with larger diameters will be necessary when using current

electrical TSV fabrication technologies. However, it is important to note that smaller diameter TSVs allow the most TSV density to be achieved, which is important for chip designs where the silicon area allowed for electrical TSV fabrication is limited.

2.5.2. Microchannel Heak Sink Geometry and Electrical TSV

Performance

When integrating microchannel heat sink technology in a 3D chip stack, cooling capability of the heat sink increases with microchannel height and wafer thickness, as shown from modeling results in the previous sections. However, thinner wafers help to facilitate easier electrical TSV fabrication. Furthermore, wafer thickness has an impact on the electrical impedances and performance of electrical TSVs.

In the following analysis, electrical properties including resistance and capacitance are modeled for electrical TSVs of various diameters (5 μ m, 10 μ m, 15 μ m, and 20 μ m). Simulation results examine the impact of electrical TSV diameter and electrical TSV length (based on wafer thickness) on TSV resistance and capacitance.

The resistance of an electrical TSV can be determined using the following equation:

$$R_{TSV} = \frac{\rho \cdot l_{TSV}}{\pi \cdot r_{TSV}^2},$$

where ρ is the resistivity of the conducting material, l_{TSV} is the length of the TSV, and . r_{TSV} is the TSV radius. In this analysis, copper is used as the electrical TSV metal.

The modeled results in Figure 2.26 and Table 2.11 show electrical TSV resistance as a function of wafer thickness and TSV diameter.



Figure 2.26: Electrical TSV resistance as a function of wafer thickness and TSV diameter.

TSV Diameter	TSV Length	TSV Resistance
(µm)	(µm)	(mΩ)
5	20	17.11
5	50	42.78
5	100	85.56
5	200	171.12
5	400	342.25
10	20	4.28
10	50	10.7
10	100	21.31
10	200	42.78
10	400	85.56
20	20	1.07
20	50	2.67
20	100	5.35
20	200	10.7
20	400	21.39

Table 2.11: Electrical TSV resistance as a function of wafer thickness and TSV diameter.
As expected, the data shows that electrical resistance is decreased for shorter TSVs and for TSVs having larger diameters.

Electrical TSVs are typically fabricated by etching holes in a silicon wafer and filling the holes with a thin layer of oxide and a metal, as shown in Figure 2.27.



Figure 2.27: Top view of an electrical TSV in silicon and the capacitances associated with the TSV structure.

The self-capacitance can be calculated by the using the following expression:

$$C_{TSV\min} = \frac{C_{OX}C_{dep\min}}{C_{OX} + C_{dep\min}},$$

where C_{OX} is the oxide capacitance and $C_{dep \min}$ is the depletion capacitance [2.18].

When the voltage applied to TSVs is higher than the flat-band voltage ($V_{TSV} \ge V_{FB}$), the substrate is depleted so that the total capacitance is the series of the oxide capacitance and the depletion capacitance (C_{dep}) [2.19].

$$C_{TSV} = \frac{1}{C_{OX}} + \frac{1}{C_{dep\min}} = \frac{C_{OX}C_{dep\min}}{C_{OX} + C_{dep\min}}$$

When capacitors are in series, the smaller capacitance dominates the value of the overall capacitance. In this case, C_{ox} results in a much smaller capacitance for most cases

considered in this work. For this analysis, the overall capacitance is estimated as being the value of C_{OX} . The TSV oxide capacitance is determined by:

$$C_{OX} = \frac{2\pi\varepsilon_{OX}L_{TSV}}{\ln(\frac{R_{via}}{R_{metal}})},$$

where ε_{ox} is the oxide permittivity, L_{TSV} is the TSV length, R_{via} and R_{metal} are the radii of the via and copper (Figure 2.27) [2.18].

The C_{ox} is modeled and plotted for different heat sink heights and different TSV aspect ratios, assuming the thickness of the dielectric (oxide) to be 1µm. The simulation results in Figure 2.28 show that the TSV capacitance increases linearly as the thickness of the die increases.



Figure 2.28: Electrical TSV oxide capacitance as a function of wafer thickness and TSV diameter.

Table 2.12 shows a summary of the simulation results. For 10 μ m diameter electrical TSVs, C_{TSV} for a 200 μ m thick wafer is 237.9fF, while the TSV capacitance for a 400 μ m thick wafer is 475.8fF. Table 2.12 also shows that TSV capacitance increases with increasing TSV diameter and increasing wafer thickness. Furthermore, as the TSV aspect ratio increases, the TSV capacitance decreases. The capacitance reduction improves the interconnect latency and power consumption.

Table 2.12: Electrical TSV capacitance as a function of wafer thickness and TSV diameter.

TSV	TSV	TSV
Diameter	Length	Capacitance
(µm)	(µm)	(fF)
1	200	39.5
1	300	59.2
1	400	79
5	200	128.9
5	300	193.4
5	400	257.8
10	200	237.9
10	300	356.8
10	400	475.8
20	200	455.1
20	300	682.6
20	400	910.1

2.6. Summary

This chapter describes the wafer-level batch fabrication and micromachining technologies that are used to fabricate the necessary electrical and microfluidic interconnects for the proposed 3D inter-layer cooling platform.

Each silicon die of the 3D stack contains the following features: 1) a monolithically integrated microchannel heat sink, 2) through-silicon fluidic vias (TSFV)

used for fluidic routing in the 3D stack, and 3) solder bumps (electrical I/Os) and microscale C4 pipes (fluidic I/Os) on the side of the chip opposite to the microchannel heat sink. Fabrication results of the individual components and their integration is demonstrated.

Additionally, compact physical modeling is used to analyze the impact of microchannel geometry and fluid flow rates on thermal resistance and pressure drop of the 3D systems. Compact physical modeling is also used to explore the electrical TSV performance and microchannel heat sink cooling trade-offs when integrating microchannel heat sinks and electrical TSVs in a 3D chip stack.

CHAPTER 3

DESIGN AND FABRICATION OF ELECTRICAL AND FLUIDIC I/O INTERCONNECTS

3.1. Integrated Electrical and Fluidic I/O Technologies for 3D Interlayer Liquid Cooling

In the 3D cooling scheme outlined in Figure 2.2, microscale fluidic interconnection between strata is enabled by through-wafer fluidic vias and fluidic I/O interconnects. Three distinct fluidic I/O technologies have been developed, including a controlled collapse chip connection (C4) pipe fluidic I/O (Figure 3.1a) [2.9], an air-gap C4 fluidic I/O (Figure 3.1b) [2.9], and a polymer pipe fluidic I/O (Figure 3.1c) [2.3-2.4].

Solder ring based fluidic sealing approaches have been previously studied to form compact 3D packages by stacking multi-layer FR-4 substrates in [3.1]. Development of a micro-scale solder-based fluidic interconnect technology allows simultaneous batch fabrication of electrical and fluidic I/Os, making electrical and fluidic I/O integration seamless. The following sections outline process integration and assembly technologies for fluidic I/O interconnect technologies which can be used to hermetically seal fluidic interfaces in 3D chip stacks.

The solder-based fluidic I/Os have multiple advantages over the polymer-based fluidic I/Os. The solder fluidic I/Os do not require polymer sockets to aid in sealing. Thus, because of the ability to fabricate shorter fluidic I/Os, there is less space between chips in the 3D stack, making the 3D stack more compact. Additionally, it is not

necessary that an epoxy-based sealant be applied at the edges of the I/Os, as the assembly of the solder fluidic I/Os to copper rings on the substrate creates a hermetic seal. Because sealant/underfill is not required for the solder-based fluidic I/Os, reworkability is possible. Also, the metal C4 I/Os have a much lower moisture absorption rate than polymer-based fluidic I/Os. Another important feature is that the fabrication of electrical and solder fluidic I/Os can be done simultaneously, and this process is compatible with existing C4 solder bumping technologies.



Figure 3.1: (a) C4 fluidic I/O, (b) air-gap C4 fluidic I/O, and (c) polymer pipe fluidic I/O.

An advantage of the polymer fluidic I/Os is the flexibility in height in which these I/Os can be fabricated. Polymer fluidic I/Os can be fabricated up to hundreds of microns in height. This I/O interconnect height flexibility could be useful when designing various 3D chip stack configurations. Table 3.1 summarizes the advantages and disadvantages of each I/O technology.

	0	vs.	0 0 0 0	
Ş	Solder Pipe	Cu Bonding Pad	Polymer Pipe	Polymer Socket
	Solder Pipe			Polymer Pipe
	No	Need for additional level of (polymer socket)	Yes	
	No	Underfill needed	Yes	
	Yes	Reworkable	No	
	Yes	Simultaneous fabrication w I/Os (fewer processing step	Νο	
	+ + +	Thermal conductivity; Ther	+	
	+ + +	Low Moisture Absorption	+	
	+	Flexibility of I/O height	+++	

Table 3.1: Comparison of fluidic I/O Interconnect technologies.

3.2. C4 Fluidic I/O Interconnects

Figure 3.2 outlines the steps necessary for fabrication of solder-based fluidic I/O interconnects. After sputtering a Ti/Cu/Ti seed layer (Figure 3.2a), photoresist is used to pattern an electroplating mold (Figure 3.2b). After electroplating a Ni under-bump metallization layer, solder is electroplated in the mold. After electroplating, the

photoresist is etched by solvent removal, the seed layer is removed by wet etching, and the solder is reflowed (Figure 3.2c).



Figure 3.2: Schematic of C4 electrical and fluidic I/O fabrication.

Figures 3.3a and 3.3b show SEM images of 30µm solder fluidic I/Os before and after reflow. The inner-diameter of the I/Os is 120µm, and the outer-diameter is 340µm. Figure 3.4 shows optical images of the solder-based fluidic I/Os.

To measure feature uniformity, the height and inner diameter of fifteen random (~25 μ m tall before and ~35 μ m tall after reflow) I/Os were measured before and after reflow. The data measured for one sample reveals that the features are verified to have good uniformity. When measuring the C4 fluidic I/Os, it was observed that the standard deviation of the height and inner diameter of the features improves after reflow. The standard deviation of the height and inner diameter of the fluidic I/Os after reflow are found to be ~0.74 μ m and ~4.16 μ m respectively (Figures 3.5a and 3.5b). Furthermore, it

is also important to note that the average height of the C4 fluidic I/Os increased $\sim 26\%$ after reflow, and the inner diameter of the C4 fluidic I/Os increased $\sim 33\%$ after reflow.



Figure 3.3: SEM images 30µm tall C4 fluidic I/Os (a.) before and (b) after reflow.



Figure 3.4: Optical images of C4 fluidic I/Os.

During the initial design of the fluidic I/Os, two shapes were considered – a circular shape (Figure 3.6a) and a square-like shape (Figure 3.6b). During electroplating, both of these feature shapes can be plated uniformly. However, the uniformity of the

features is best characterized after solder reflow. In the square-like I/O structure, the sides and the edge of the feature reflow to different heights (Table 3.2). Yet, the circular pipe-like structure has uniform height after reflow (Table 3.3).



Figure 3.5a: C4 Fluidic I/O height uniformity before and after reflow.



Figure 3.5b: C4 Fluidic I/O inner diameter uniformity before and after reflow.



Figure 3.6 Feature Height Uniformity Shape Dependence – (a) circular solder fluidic I/O and (b) square-like solder fluidic I/O.

Table 3.2: Characterization of feature height of square-like fluidic I/O.

Height	Before Reflow	After Reflow
Side	33µm	37µm
Edge	34µm	46µm

Table 3.3: Characterization of feature height of circular fluidic I/O.

Height	Before	After	
	33.5µm	45µm	

3.3. Integrated C4 Electrical and Fluidic I/O Interconnects

When using the solder-based fluidic I/O technologies, electrical and fluidic I/Os can be fabricated using a single masking step. The fabrication process yields high density electrical I/Os (\sim 1600/cm²) which have a pitch of 240µm, although smaller pitches are possible. The adjacent fluidic I/Os have a pitch of 480µm (Figure 3.7). The fabrication of electrical and fluidic I/Os yields features with good uniformity and a

standard deviation in feature size of $<1\mu$ m. Figures 3.8a and 3.8b show SEM images of integrated electrical and fluidic I/O interconnects before reflow and after reflow.



Figure 3.7: Schematic of the chip layout design for C4 electrical and fluidic I/Os.



Figure 3.8a: SEM image of integrated C4 electrical and fluidic I/Os before reflow.



Figure 3.8b: SEM image of integrated C4 electrical and fluidic I/Os after reflow.

For ideal assembly conditions, it is preferred that the height of the solder bumps be slightly taller than the height of the solder pipes in order to ensure electrical connectivity during flip-chip assembly. Thus, ten different C4 bump-pipe feature size combinations were fabricated in order to achieve a feature size combine that resulted in a 2µm height difference between the solder bumps and the solder pipes.

Experimental results revealed that as the surface area of the solder bump and fluidic I/O become closer in value, the post-reflow feature height becomes almost equal in height. Furthermore, it is also important to note that the "solder pipe thickness" (ring thickness) is similar to the diameter of the solder bump for feature combinations that that have a similar post-reflow feature height. For example, for a 340µm outer diameter (140µm inner diameter) pipe, the pipe/ring thickness is 100µm. Similarly, the bump diameter is also 100µm. Table 3.4 outlines the bump-pipe feature sizes that yielded the best results for obtaining the desired C4 electrical and fluidic I/O post-reflow height.

Wafer 1	Before Reflow	After Reflow	σ	Bump Dia	80µm
Bump	30µm	38.5µm	<1µm	Pipe Dia	300µm
Pipe	31.5µm	40.5µm	<1µm	Fluidic TSV Dia	100µm
Wafer 2	Before Reflow	After Reflow	σ	Bump Dia	100µm
Bump	32µm	43µm	<1µm	Pipe Dia	340µm
Pipe	33.5µm	45µm	<1µm	Fluidic TSV Dia	100µm
Wafer 3	Before Reflow	After Reflow	σ	Bump Dia	120µm
Bump	34µm	47.5µm	<1µm	Pipe Dia	340µm
Pipe	36µm	45.5µm	<1µm	Fluidic TSV Dia	100µm

Table 3.4: Feature sizes of various bump-pipe combinations before and after solder reflow.

3.3.1. Integrated C4 Electrical and Rectangular Fluidic I/O

Interconnects

A rectangular fluidic I/O design was also considered. This fluidic I/O design requires the fabrication of a single square-like fluidic via and fluidic I/O on each side of the chip. Figure 3.9 shows SEM images of integrated C4 electrical and rectangular fluidic I/Os.



Figure 3.9: SEM images of integrated C4 electrical and rectangular C4 fluidic I/Os.

An advantage of using the rectangular fluidic I/O design is that the pressure drop across the microfluidic system is decreased, as the cross sectional area of the opening for a singular rectangular fluidic I/O is much larger than the area of the smaller individual circular fluidic I/Os. However, a very important advantage of using the circular C4 fluidic I/O is that the circular fluidic I/Os require less silicon area. Additionally, using individual circular I/Os instead of a single fluidic I/O that spans almost the entire length of the chip uses less silicon area near the edges of the chip that may be required for the fan out of signal interconnects from the electrical I/Os.

Table 3.5 outlines a summary of the equations that can be used to calculate the pressure drop and silicon area occupied for each fluidic I/O design [2.8,3.3]. A comparison of the pressure drop of each fluidic I/O design is summarized in Table 3.6, and the silicon area occupied by each fluidic I/O design is summarized in Table 3.7.

The pressure drop and silicon area calculations for the circular fluidic I/Os include the following assumptions: chip area is 1cm x 1cm, number of fluidic I/Os and fluidic vias = 17, fluidic via and fluidic I/O diameter = 100um, height of fluidic via = 400um, height of fluidic I/O = 50um. The pressure drop of a rectangular fluidic I/O can be calculated by using the equation in Table 3.5 Figure 2.14. The pressure drop and silicon area calculations for the rectangular fluidic I/Os include the following assumptions: chip area is 1cm x 1cm, number of fluidic I/Os and fluidic vias = 2, fluidic via and fluidic I/O width = 100um, height of fluidic via = 400um, height of fluidic I/O = 50um.

Analysis shows that, for a 1 cm^2 chip, the pressure drop when using 26 circular fluide I/Os is ~3.4kPa compared to a pressure drop of only 0.42kPa when using the 2 rectangular fluidic I/Os. However, although the rectangular fluidic I/Os result in a lower

pressure drop, the rectangular fluidic I/Os occupy 70% more silicon area than the circular fluidic I/Os. Consequently, the tradeoff of pressure drop and silicon area must be considered when deciding which fluidic I/O design to implement.

Equations	Symbol Legend
$\Delta P_{via+micropipe} = 512 \cdot \frac{\mu \cdot V \cdot (2H_{via} + 2H_{pipe})}{n_{via} \cdot \pi \cdot D_{via}^4}$	D_{via} = fluidic via diameter $\Delta P_{via+squareIO}$ = pressure drop in circular fluidic I/Os and fluidic vias
$\Delta P_{via+squareIO} = \frac{1}{2} f \cdot \text{Re} \cdot L_{SqIO} \cdot \frac{\mu \cdot V(1 + W_{SqIO} / H_{SqIO})^2}{n_{via} \cdot H_{SqIO} \cdot W_{SqIO}^3}$	$\Delta P_{via+micropipe}$ = pressure drop in rectangular fluidic I/Os and fluidic vias
$H_{SqIO} = H_{via} + H_{SqIO}$	f = friction coefficient Re = Reynolds number
Silicon area for fluidic IOs	V = flow rate
$\frac{bincont area}{Chip} = \% area occupied by$	n_{vias} = number of fluidic vias
fluidic IOs	μ = fluid kinematic viscosity
	H = height
	L = length
	W = width

Table 3.5: Hydro-dynamic equations for calculating pressure drop in fluidic I/Os [2.8, 3.2].

Fluidic I/O Design	Number of Fludic Vias per Chip	Number of Fludic I/Os Per Chip	Fluidic I/O Height (µm)	Fluidic I/O Width (inner) (µm)	Fluidic I/O Length (cm)	Fluidic I/O Diameter (µm)	Wafer Thickness (µm)	Pressure Drop (kPa)
Circular	34	34	50	n/a	n/a	100	400	3.4
Rectangular	2	2	50	100	1	n/a	400	0.42

Table 3.6: Summary of pressure drop calculations for fluidic I/O designs.

Table 3.7: Summary of chip area occupied by fluidic I/Os.

Fluidic I/O Design	Number of Fludic Vias per Chip	Number of Fludic I/Os Per Chip	Fluidic I/O Width (outer) (µm)	Fluidic I/O Length (cm)	Fluidic I/O Outer Diameter (µm)	Area (cm ²)	% Chip Area Occupied
Circular	34	34	n/a	n/a	340	0.03	~ 3%
Rectangular	2	2	500	1	n/a	0.1	~ 10%

3.4 Air-gap C4 Fluidic I/O Interconnects

Air-gap C4 I/Os can be fabricated in a similar manner as the circular C4 fluidic I/Os in the previous section. The air-gap fluidic I/Os were made by electroplating solder in the fluidic I/O mold to be ~twice the height of the mold. So, for a 25μ m tall mold, it would be necessary to electroplate ~50µm of solder to achieve an air-gap in the structure.

To fabricate a taller structure, 50 μ m of solder was electroplated in a 25 μ m photoresist electroplating mold. As the height of the electroplating mold was 25 μ m, the solder was over-plated by a height of 25 μ m to form a 50 μ m tall solder pipe I/O structure. Additionally, when over-plating solder (i.e., height of plated solder is greater than the height of the resist mold), the inner diameter of the electroplated structures decreases to ~25 μ m (Figure 3.10a). After reflow, because of the decreased inner-diameter of the structure, the solder at the top of the structure merges (Figure 3.10b). Although, these

reflowed fluidic I/Os appear to look like solder bumps, the domed structures actually have an air-filled depression in the middle of the structure.

This air-filled depression occurs because when solder begins to electroplate over the height of the photoresist electroplating mold, the over-plated solder decreases the C4 fluidic I/O inner diameter only at the top of the electroplated fluidic I/O. The inner diameter on the bottom half of the air-gap C4 fluidic I/O cannot become smaller prior to reflow because the solder at the bottom half of the fluidic I/O is still separated by photoresist mold. After the photoresist is removed and when reflow occurs, the solder merges only at the top of the fluidic I/O, where the inner diameter of the pipe structure has become smaller.



Figure 3.10: 50µm tall, over-plated air-gap C4 fluidic I/Os (a) before and (b) after reflow.

The air void is verified by the x-ray image of the structure taken after reflow (Figure 3.11). All visually inspected bumps contained the air-gap. One key advantage to

the air-gap fluidic I/O technology is that it is transparent to the flip-chip assembly process. Furthermore, the air-gap C4 technology enables the ability to use no-flow underfill in applications for which it is required.



Figure 3.11: X-ray image of air-gap C4 fluidic I/Os after solder reflow.

3.5. Integrated Polymer Sockets, Fluidic TSVs, Electrical I/Os, and

Polymer Pipe Fluidic I/Os

An alternative fluidic sealing approach can be implemented using polymer fluidic I/Os. Figure 3.12 shows a schematic of the fabrication process flow for a silicon die with integrated polymer sockets, through-wafer fluidic interconnects, thermofluidic I/O interconnects, and electrical I/O interconnects. The process begins by sputtering a

300/10000/300Å titanium/ copper/titanium (Ti/Cu/Ti) metal layer, where Ti serves as an adhesion promoter between Cu and silicon (Figure 3.12a). The metal is patterned using a wet etch process (Figure 3.12b). Next, 1µm of oxide is deposited on the back side of the wafer as a polymer adhesion layer (Figure 3.12c), and 3µm of oxide is deposited on the front side as a through-silicon via etch-stop layer (Figure 3.12d). Next, 15µm of Avatrel 2090P polymer is spin coated onto the wafer (Figure 3.12e). Afterwards, polymer sockets are patterned on top of the metal (Figure 3.12f, Figure 3.13). The first layer of Ti is removed using a wet-etching process. Through-wafer fluidic vias are patterned and anisotropically etched into the back side of the silicon wafer in an ICP etching tool (Figure 3.12g, Figure 3.14); the etching stops at the etch-stop layer on the front side of the wafer. Next, a 12µm layer of Avatrel polymer is spin coated and patterned on the front side of the wafer and used as a passivation layer (Figure 3.12h). After sputtering a 300/2000/300Å Ti/Cu/Ti seed layer and electroplating a 2µm nickel under-bump metallurgy layer, 50µm C4 solder bumps are electroplated for area-array electrical interconnects (Figure 3.12i, Figure 3.15). Afterwards, a 60µm layer of Avatrel polymer is spin coated onto the front side of the wafer and used to pattern polymer pipes, which serve as thermofluidic I/O interconnects (Figure 3.12j, Figure 3.16). Finally, the oxide layer covering the through-wafer fluidic vias on the front side of the wafer is removed using a wet etch process to allow fluidic circulation.

As outlined in Table 3.3, an advantage of the polymer based fluidic I/Os is the flexibility in height in which these I/Os can be fabricated. Polymer-based I/Os can be fabricated up to hundreds of microns in height. This I/O interconnect height flexibility could be useful if one were to design a processor memory chip stack in which fluidic I/Os

were needed to route liquid to high-performance processor chips separated by memory chips in the 3D chip stack. The ability to fabricate longer I/Os would enable routing of liquid over longer distances in a multi-chip 3D stack, as shown in Figure 3.17.



Figure 3.12: Schematic of wafer-level integration of polymer sockets, electrical I/O and fluidic I/O interconnects for a 3D flip-chip package.



Figure 3.13: SEM image of 15µm tall, 270µm and 60µm diameter polymer sockets.



Figure 3.15: SEM image of 50µm tall solder bumps used as electrical I/O interconnects.



Figure 3.14: SEM image of 100µm diameter through-wafer fluidic interconnects.



Figure 3.16: SEM image of 60µm tall polymer pipes used as I/O fluidic interconnects.



Figure 3.17: Schematic illustrating how polymer fluidic I/Os can be used to route liquid over longer distances in the 3D chip stack [3.3].

3.6. Summary

Chapter 3 describes the fabrication and process integration techniques for three distinct fluidic I/O technologies including a C4 pipe fluidic I/O, an air-gap C4 fluidic I/O, and a polymer pipe fluidic I/O interconnect technology. An electrical I/O density of \sim 1600/cm² is demonstrated. The advantages and disadvantages of the three fluidic I/O technologies are discussed.

CHAPTER 4

3D CHIP ASSEMBLY USING ELECTRICAL AND FLUIDIC I/O INTERCONNECTS

4.1 3D Chip Stacking Using Electrical and Fluidic I/O Interconnects

A challenge in such a 3D configuration, as shown in Figure 2.2, is the flip-chip bonding process, especially since one must be able to provide fluidic sealing to prevent leakage in the 3D chip stack. Because there are three distinct fluidic I/O technologies, different 3D stacking and assembly methods are required for the assembling microfluidic chips in the 3D chip stack, depending on which fluidic I/O technology is implemented. This section discusses the flip-chip bonding processes that enable assembly of the microfluidic chips in the liquid-cooled 3D chip stack.

4.2 Assembly of C4 Electrical and C4 Fluidic I/Os

After fabrication of the electrical and fluidic I/O interconnections, as shown in Figure 2.3, the solder-based electrical and fluidic I/Os can be simultaneously aligned and assembled to a substrate which has patterned copper pads and copper rings. The solder bumps (electrical I/Os) are assembled to the copper pads, and the solder pipes (fluidic I/Os) are assembled to the copper rings, as shown in Figure 4.1.

The copper pads and copper rings on the substrate have slightly larger features sizes than the C4 electrical and fluidic I/Os on the chip (Figure 4.1). After copper is deposited on the substrate, the copper pads and rings can be patterned by silicon dioxide

or polymer, which is needed to contain the solder during assembly, as shown in Figure 4.2. Patterning the copper rings with a polymer layer of an appropriate thickness can provide an additional level of fluidic sealing. SEM images of the copper pads and copper rings on the substrate are shown in Figure 4.3.



Figure 4.1: Characterization and assembly of C4 fluidic I/Os.

When the C4 fluidic I/Os are assembled to the substrate, the solder only wets the patterned copper on the substrate. The C4 electrical and fluidic I/Os and the copper pads and rings on the substrate are simultaneously aligned and brought in to contact with an appropriate force and temperature. Subsequently, an electrical and mechanical connection is formed between the solder bumps and copper pads, and a hermetically sealed fluidic pathway is created when the solder pipes are bonded to the corresponding copper pads on the substrate (Figure 4.2).



Figure 4.2: Schematic of assembly of C4 electrical and fluidic I/Os.

For ideal assembly conditions, it is preferred that the height of the electrical I/Os be slightly taller than the height of the fluidic I/Os in order to ensure electrical connectivity. Thus, various solder bump diameters were fabricated in order to achieve a 2μ m height difference between the solder bumps and the solder pipes. Table 4.1 outlines the feature heights that were fabricated when producing the subsequent assembly results.



Figure 4.3: SEM image of substrate with patterned copper rings (360µm outer diameter) and integrated fluidic TSVs (left) and an SEM image integrated copper pads and copper rings.

Height	Before	After
Bump	34µm	47.5µm
Pipe	36µm	45.5µm

Table 4.1: Before and after reflow feature height parameters for assembly.

A Finetech Fineplacer Lambda flip-chip bonder was used to perform assembly of the 3D chip stack. Figure 4.4 shows a photograph of the Finetech Fineplacer Lambda flip-chip bonder, which has an alignment accuracy of 0.5µm.



Figure 4.4: Photograph of Finetech Fineplacer Lambda flip-chip bonder.

The process used for assembly is summarized in Table 4.2, and the bonding temperature profile used for assembly is shown by the reflow profile image that is generated by the Finetech bonding during assembly (Figure 4.5). After pre-heating the die and the substrate to temperatures of 60°C and 40°C, respectively, the two are brought into contact with a compression force of 100g. Subsequently, the temperature of the chip

and the substrate are elevated to 235°C and 140°C, respectively. The 340µm diameter airgap C4 fluidic I/Os are aligned to the 360µm diameter copper rings on the substrate (Figure 4.3).

Assembly Parameters and Sequence of Steps	Value
Pre-heating Temperature of Substrate	40°C
Pre-heating Temperature of Die 1	60°C
Compression Force	100g
Bonding Temperature of Substrate	140°C
Bonding Temperature of Die 1	235°C

Table 4.2 Bonding process parameters.



Figure 4.5: Bonding process parameters.

4.3 Assembly of C4 Electrical I/Os and Air-gap C4 Fluidic I/Os

The assembly process for C4 electrical I/Os and air-gap C4 fluidic I/Os is similar to flip-chip bonding process shown in the previous section. When the air-gap C4 fluidic

I/Os are assembled to the substrate, the solder only wets the patterned copper on the substrate. Thus, the assembly of the air-gap C4 fluidic I/Os to the copper rings on the substrate enables the air-filled solder bump-like structures to be shaped into solder pipe-like fluidic I/O interconnects. The assembly process is similar for the C4 fluidic I/Os. However, since the fluidic pipe structures are already formed, the hermetically sealed fluidic pathway is created when the solder pipe is bonded to the copper pads, as shown in Figure 4.6 and Figure 4.7.



Figure 4.6: Characterization and assembly of air-gap C4 fluidic I/Os.



Figure 4.7: 3D assembly using air-gap C4 fluidic I/Os and electrical I/Os.

For the air-gap C4 fluidic I/O, the formation of the C4 fluidic pipe structure is verified by x-ray images taken of the I/Os after assembly (Figure 4.8, Figure 4.9). Figure 4.10 shows a SEM image of a 2-chip stack of microfluidic chips with C4 electrical and fluidic I/Os after assembly.



Figure 4.8: X-ray image of air-gap C4 fluidic I/Os after assembly.



Figure 4.9: X-ray image of air-gap C4 fluidic I/Os after assembly.



Figure 4.10: SEM image of a 3D chip stack assembled using C4 fluidic and electrical I/Os.

It is important to note the significance of optimizing the appropriate bonding force and having a sufficiently large C4 fluidic I/O inner diameter, which must be larger than the diameter of the fluidic TSV. Experimental results revealed that, for chips that contained 100µm diameter fluidic TSVs, fluidic TSV clogging was observed when the inner diameter of the C4 fluidic I/Os were fabricated to be less than 130µm. Furthermore, for a chip with an area of 0.7cm x 0.7cm, some fluidic I/O bonding sites across the chip were observed to have fluidic TSV clogging when flip-chip assembly was performed at a bonding force over 100g. Figure 4.11 shows a cross-sectional optical image of an assembled 3D chip stack where solder from the fluidic I/Os is clogging fluidic vias in the chip stack. Figure 4.12 shows a cross-sectional optical image of a 3D chip stack assembled using C4 fluidic and electrical I/Os. An optimal amount of bonding force and an appropriate amount of fluidic TSV clearance were used to prevent fluidic TSV clogging, and a clear fluidic pathway is created for facilitating fluid flow to each layer in the 3D chip stack.



Figure 4.11: Optical image showing fluidic TSV clogged with solder.



Figure 4.12: Cross-sectional optical image of a 3D chip stack assembled using C4 fluidic and electrical I/Os.

As discussed in Chapter 3, a rectangular fluidic I/O configuration can be used to reduce the pressure drop associated with the chip fluidic I/Os. The same flip-chip assembly recipes and principles used for assembly of the C4 electrical and fluidic I/Os can be applied to performing assembly of C4 electrical I/Os and C4 rectangular fluidic I/Os. Figure 4.13a presents a SEM image of integrated C4 electrical and rectangular fluidic I/Os, and Figure 4.13b shows an optical image of the corresponding copper pads that are fabricated on the substrate to facilitate assembly. Figure 4.14 shows x-ray images of successful 3D chip stacking using C4 electrical and rectangular fluidic I/Os.



Figure 4.13: (a.) Integrated C4 electrical and rectangular C4 fluidic I/Os. (b.) Copper pads fabricated on the substrate to facilitate assembly.



Figure 4.14: X-ray images of successfully bonded 3D stack of chips with C4 electrical and rectangular fluidic I/Os.

4.4 Fluidic Testing of C4 Fluidic I/O Interconnects

To test the C4 fluidic I/Os, two chips were assembled to a silicon substrate. A mechanical pump and fluid inlet pipe were attached to the bottom side of the silicon substrate in the 3D stack and used to pass fluid through the chip stack (Figure 4.15). It was observed that fluid can be delivered from the bottom of the stack, through the 3D stack, and out of the fluidic vias of the top-most chip. Fluid was circulated through the fluidic I/Os at flow rates up to 100ml/min, and no leakage was observed. As previously discussed in Chapter 2, for the microchannel heat sink configuration used in the

experimental component of this work, a fluid flow rate of 100ml/min would be sufficient for achieving a desired thermal resistance of >0.28 °C/W.



Figure 4.15: Experimental setup for fluidic testing of C4 fluidic I/Os.

4.5 Assembly Using Polymer Sockets, Polymer Pipe Fluidic I/Os, and Electrical I/Os

Microfluidic chips can also be stacked when using polymer pipe fluidic I/Os and solder electrical I/Os. To develop the necessary assembly processes, a silicon chip was fabricated which contained the following features: fluidic TSVs, polymer pipe I/O interconnects, high density solder bump electrical I/O interconnects on the front side of the dice, and polymer sockets on the back side of the dice, as shown in Figure 4.16. The silicon substrate contains copper pads, polymer sockets, and integrated fluidic TSVs.



Figure 4.16: 3D assembly using polymer fluidic I/Os and solder electrical I/Os.

Using this configuration, dice were aligned, stacked, and assembled on a silicon substrate using a RD Automation flip-chip bonder that has an alignment accuracy of $<2\mu$ m. The process used for assembly involves pre-heating the die and the substrate to temperatures of 180°C and 140°C, respectively, bringing the two into contact with a compression force of 200g, and elevating the temperature of the chip and the substrate to 230°C and 150°C, respectively. The bonding process parameters are listed in Table 4.3.

The fluidic I/Os and electrical I/Os are assembled simultaneously. The 250 μ m diameter polymer pipes are aligned to the 270 μ m diameter polymer sockets on the substrate, and the 50 μ m diameter solder bumps are aligned to the copper traces and 60 μ m polymer sockets on the substrate (Figure 4.16). In addition to serving as electrical and
fluidic I/Os, the solder bumps and polymer pipes provide mechanical interconnection between the bottom die and the substrate and between the dice in the 3D stack.

Because copper pads and polymer sockets are fabricated onto the back side of the first die, the second die can assembled onto the back side of the first die using the same bonding recipe. The self-alignment property of solder increases the alignment accuracy of the die to the substrate. Patterned silicon dioxide on the substrate contains the solder during reflow. The process used for assembly of the 3D prototype is thus compatible with conventional flip-chip bonding. Figure 4.17 shows an SEM cross-sectional image of a 3D stack of two microfluidic chips. Figures 4.18a and 4.18b show infrared microscope images of through-silicon fluidic via alignment of the two chips. Figure 4.19 shows a 3D stack of two chips assembled to a silicon substrate and a 3D stack of four chips assembled to a silicon substrate.

Assembly Parameters and Sequence of Steps	Value
Pre-heating Temperature of Substrate	140°C
Pre-heating Temperature of Die 1	180°C
Compression Force	200g
Bonding Temperature of Substrate	150°C
Bonding Temperature of Die 1	230°C
Pre-heating Temperature of Substrate and Die 1	140°C
Pre-heating Temperature of Die 2	180°C
Compression Force	200g
Bonding Temperature of Substrate and Die 1	180°C
Bonding Temperature of Die 2	230°C

Table 4.3: Bonding Process Parameters.



Figure 4.17: Cross-sectional SEM image of 3D microfludic chip-to-chip bonding.



Figure 4.18a: Top-view IR-microscope image of 3D stack. Figure 4.18b: Tilted IR-microscope image of 3D stack.

After assembly, to seal the fluidic I/Os on the front side of each chip, an epoxybased underfill is applied at the edges of chip (Figure 4.20). The underfill provides a stronger mechanical connection between each interface. For this application, most importantly, underfill is used for the purpose of sealing the fluidic interconnect interfaces between the die and the substrate and between the dice in the 3D stack.

To test the reliability of the fluidic sealant, a syringe pump and fluid inlet pipe were attached to the back side of the top-most die in the 3D stack and used to pass fluid through the chip stack (Figure 4.21). Fluid can be delivered from the top chip, through the 3D stack, and out of the bottom of the substrate with no leakage at the chip-to-chip and chip-to-substrate interfaces. Consequently, based on this preliminary test, this fluidic I/O technology can be used to route fluid to each layer of microchannels in a 3D chip stack.



Figure 4.19: SEM images of 2-chip and 4-chip 3D stacks.



Figure 4.20: 3D assembly and fluidic sealing.



Figure 4.21: Experimental setup for fluidic testing of polymer pipe fluidic I/Os.

4.6. Summary

Chapter 4 discusses the flip-chip bonding processes that enable assembly of the microfluidic chips in the liquid-cooled 3D chip stack. Bonding process parameters are outlined for the flip-chip die-to-substrate and die-to-die bonding processes for three fluidic I/O sealing technologies.

The ability to assemble chips with integrated electrical and fluidic I/Os and seal fluidic interconnections at each strata interface is demonstrated using three assembly and fluidic sealing techniques. Assembly results show the stacking of up to four chips that contain integrated electrical and fluidic I/O interconnects, with an electrical I/O density of $\sim 1600/\text{cm}^2$. Fluidic testing is performed by circulating fluid through the 3D stack at flow rates up to 100ml/min, with no fluid leakage observed.

CHAPTER 5

THERMAL AND FLUIDIC TESTING OF THE 3D INTER-LAYER LIQUID COOLING PLATFORM

5.1 Experimental Thermal Test-bed for 3D Liquid Cooling

An experimental thermal test-bed for evaluating the cooling needs for 3D chip stacks which contain high-performance microprocessors has been developed. This testbed enables the thermal simulation and experimental thermal measurements of liquidcooled processor-on-processor (Figure 5.1) and memory-on-processor (Figure 5.2) 3D chip stacks.



Figure 5.1: Schematic of processor-on-processor 3D chip stack.



Figure 5.2: Schematic of memory-on-processor 3D chip stack.

The controllable parameters of the experimental setup include power dissipation and heat flux of each chip in the 3D stack, fluid flow rate, the number of I/Os on a chip, and selective layer heating and cooling. The measurable parameters of the chips in the 3D stack include thermal resistance of each chip in the stack, average pressure drop of the 3D stack, and temperature rise of each chip.

5.2 Fabrication of Top Layer in 3D Chip Stack

Fabrication of the top layer of the 3D stack, chips which contain a microchannel heat sink, fluidic through silicon vias, fluidic I/Os, electrical I/Os, and integrated platinum heaters/resistors (used for heating and temperature sensing), has been demonstrated. Figure 5.3 shows the design layout schematic of the top layer in the 3D stack, and Figure 5.4 outlines the fabrication process flow for fabrication of the top layer of the 3D chip stack.



Figure 5.3: Schematic of the top view of mask design with integrated (top layer).

A summary of the fabrication process for the chip on the top layer includes fabrication of the platinum resistor network (Figure 5.4a), etching of microchannels and fluidic TSVs in the wafer (Figure 5.4b), capping the microchannels with a silicon capping wafer (Figure 5.4c), and fabrication of copper pads, electrical I/Os, and fluidic I/Os (Figure 5.4d). To verify the depth of the microchannels, the Veeco Wyko profilometer, which uses the phase change of light reflecting from various heights of similar materials to measure the uniformity of a flat surface or the horizontal distance between two adjacent surfaces, was used. The channel depth measurement shown in Figure 5.5 verifies the depth of the microchannels to be ~350µm, which is a sufficient channel depth for obtaining the desired microchannel heat sink thermal resistance, as discussed in Chapter 2.



Figure 5.4: Schematic of wafer-level integration of microchannels, fluidic through-silicon vias, electrical and fluidic I/Os, and integrated platinum heaters/temperature sensors (top layer).



Figure 5.5: Veeco Wycko microscope measurement of microchannel depth.

5.3. Fabrication of Bottom Layer in 3D Chip Stack

Figure 5.6 shows the design layout schematic of the top layer in the 3D stack, and Figure 5.7 shows a photograph of the bottom layer in the 3D stack. The bottom side of this layer contains thin-film platinum resistors for heating and temperature sensing. The bottom side of this layer also contains copper pads, which enable power to be supplied to the heaters. This layer also contains a silicon-capped microchannel heat sink with integrated fluidic TSVs in the silicon cap. The top side of this layer contains copper pads which are used as a bonding site for the next chip in the 3D stack. Additionally, four larger copper electrode pads are present on the top side for supplying power to the thinfilm platinum heaters of the next chip in the 3D stack.



Figure 5.6: Schematic of top view of mask design with integrated (bottom chip).



Figure 5.7: Photograph of the bottom layer in the 3D stack.

Figure 5.8 outlines the fabrication process flow for fabrication of the top layer of the 3D chip stack. A summary of the fabrication process for the chip on the bottom layer includes fabrication of the platinum resistor network (Figure 5.8a), etching of microchannels and fluidic inlet and outlet channels in the wafer (Figure 5.8b), capping

the microchannels with a silicon capping wafer (Figure 5.8c), and fabrication of copper pads on the top and bottom sides of the wafer (Figure 5.8d), deposition and patterning of oxide and copper bonding pads on the top side that enables assembly of the next chip in the 3D stack (Figure 5.8e), deposition and patterning of a dielectric layer (solder or polymer) for containing solder during flip-chip assembly, and etching the silicon capping layer to expose the fluidic inlet and outlet ports that are used for supplying fluid to the 3D chip stack (Figure 5.4g).



Figure 5.8: Schematic of wafer-level integration of microchannels, fluidic TSVs, fluidic inlets/outlets, platinum heaters, and copper bonding pads (bottom layer).

Figure 5.9 shows a SEM image of the microchannels heat sink that is fabricated in the bottom silicon capped layer. The larger fluidic channels on the left and right sides of the microchannel heat sink are used for routing liquid to the microchannels on the bottom layer and ultimately to the entire 3D chip stack.



Figure 5.9: SEM image of microchannels in the 3D chip stack bottom layer, where inlet and outlet channels are on the left and right sides of the microchannel heat sink (middle).

5.4. Fluidic Connectivity within the 3D Chip Stack

Figure 5.10 shows a photograph of the multi-layer 3D stack. The features of the assembled microfluidic chip includes a silicon-capped microchannel heat sink, integrated fluidic TSVs, electrical and fluidic I/Os on the bottom side of the chip, and thin-film platinum heaters on the bottom side of the chip. "Nanoports" (Upchurch Scientific, Inc.) are used to attach fluidic tubing to the fluidic inlet and outlet and are bonded to the 3D stack, as shown in Figure 5.10, Figure 5.11, and Figure 5.12. Thus, cooling fluid can be pumped to the fluidic nanoport inlet on the bottom layer, circulated through each chip in the 3D stack, and pumped out of the fluidic nanoport outlet.

Figure 5.11 shows a schematic of a fluidic tube that originates from a mechanical pump and a liquid reservoir. Thus, cooling fluid can be pumped to the fluidic nanoport inlet on the silicon carrier bottom layer, circulated through each chip in the 3D stack, and pumped out of the fluidic nanoport outlet into a liquid reservoir.

Fluidic continuity and fluidic sealing for the 3D microfluidic network was verified and demonstrated by stacking two microfluidic chip layers, as shown in the configuration shown in Figure 5.11. After assembling the chips in the 3D stack, DI water was circulated through the 3D stack for 1 hour at flow rates ranging from 10 ml/min to 100 ml/min.



Figure 5.10: Photograph of the multi-layer 3D stack.



Figure 5.11: Schematic representation of the multi-layer 3D stack.



Figure 5.12: Photograph of nanoports and fluidic tubing attached to the 3D stack for fluidic continuity testing.

5.5. Platinum Heater / Thermometer Design and Fabrication

Thin-film platinum resistors are fabricated on the bottom side of each chip of the 3D stack (Figure 5.13). When applying a current source to the resistors, the resistors serve as heating sources which simulate heat dissipated by transistors and interconnects on a microprocessor. The change in resistance of the resistor can be measured and used to calculate the change in chip temperature. A schematic of the mask layout for the platinum heater structure is shown in Figure 5.14, and the characteristics of the platinum resistors/heaters can be found in Table 5.1. The platinum heaters that are integrated on the bottom side of each chip were designed in order to be able to dissipate a heat flux equivalent to 100W/cm².



Figure 5.13: SEM images of (left) microchannel heat sink and (right) integrated thin-film platinum resistors and electrical I/Os.



Figure 5.14: SEM images of (left) microchannel heat sink and (right) integrated thin-film platinum resistors and electrical I/Os.

Change of DC resistance of the platinum heaters was verified to be a linear function of the temperature. To characterize the temperature dependence of the Ti/Pt resistors, the sample was heated to various temperatures in a small oven and the DC resistance of the individual heaters was then measured. Figure 5.15a and Figure 5.15b show the measurement results that verify the linear relation between the heater resistance

and the temperature. The slope of the R-T curves is approximately $1.8 - 2\Omega/^{\circ}C$, which yields sufficient sensitivity for resistance measurement using a digital multimeter. Good consistence was also observed when testing different chips with integrated platinum heaters fabricated on the chip.



Figure 5.15a: Change of DC resistance of the heater as a linear function of temperature (ACHS).



Figure 5.15b: Change of DC resistance of the heater as a linear function of temperature (MCHS).

Heater Dimensions				
Heating Area	Wire Resistance	Applied Current	Applied Power	Power Density
0.36 cm ²	36 ohms	1 A	36 W	$100 W/cm^2$

Table 5.1: Thin-film platinum heater characterization.

5.6. Electrical Connectivity in the 3D Chip Stack

The ability to achieve simultaneous electrical connectivity between multiple layers in the 3D stack was demonstrated by stacking chips that contain platinum resistors, electrical I/Os, and copper pads, as shown in Figure 5.16. After assembly, resistance measurements were taken by probing the copper pad electrodes on each layer in the 3D stack. The resistance measurement of the top layer in the stack was measured to be approximately 37.1Ω , which is approximately the desired resistance measurement for the calculated resistor wire resistance shown in Table 5.1.



Figure 5.16: Experiential setup for measuring multi-layer electrical connectivity.

5.7. Experiential Thermal Measurement Comparison of an Air-cooled Heat Sink vs. a Microchannel Heat Sink

5.7.1. Experiential Thermal and Fluidic Testing Setup

The equipment required for thermal-fluidic testing and characterization includes a mechanical liquid pump, a digital flow meter for monitoring liquid flow rates, a differential pressure gauge for measuring pressure drop, thermocouples to measure the inlet and outlet temperatures of the cooling liquid, and a data logger to collect data during

thermal cycling of the chips in the 3D chip stack. Figure 5.17 and Figure 5.18 show a schematic and photograph of the experiential setup.



Figure 5.17: Schematic of the thermal test-bed experiential setup.



Figure 5.18: Photograph of the thermal test-bed experiential setup.

Microfluidic chips containing all of the components that were described in the previous sections were connected to the experimental setup to enable thermal and fluidic measurements. Additionally, a chip containing a platinum resistor network identical to the resistor network on the microfluidic chips was connected to an air-cooling heat sink (ACHS).



Figure 5.19: Schematic of ACHS and MCHS thermal testing setup.

5.7.2. Heat Sink Volume Comparison for Air-cooling vs. Microchannel Liquid Cooling

By placing the MCHS sample and the ACHS sample side-by-side, we can clearly see the volume difference of the two cooling technologies. For comparison, the dimensions of the MCHS are $(0.6 \times 0.6 \times 0.02)$ cm, while the dimensions of the ACHS are $(13 \times 10 \times 6)$ cm. The volume of the two heat sinks differs by a factor of 10^5 , which is rather significant. The volume of the two heat sink technologies differs by a factor of 10^5 . Using MCHS technology enables the reduction of cooling hardware size from inches to microns.



Figure 5.20: Heat sink volume comparison between the Intel i7 air-cooling heat sink and the microchannel heat sink.

Table 5.2: Heat sink volume comparison between the Intel i7 air-cooling heat sink and the microchannel heat sink.

Technology	Volume	
Air-cooling Heat Sink	(13 x 10 x 6) cm ³	
Microchannel Heat Sink	(0.6 x 0.6 x 0.03) cm ³	

5.7.3. Thermal Resistance Analysis for Air-cooled Heat Sink vs.

Microchannel Heat Sink

In the ACHS experiment, we use a commercially available CPU cooler which consists of 3 copper heat pipes and 45 aluminum fins that is designed for the Intel i5/i7 CPU. The ACHS sample is tested while the fan is rotating at its maximum speed (2500rpm ± 15). The corresponding air flow is 54.8 CFM.

For the MFHS testing, we used the sample with an embedded micropin-fin heat sink, which is fabricated as described in the previous section. The thermal measurements are made at two flow rates: 50 ml/min and 75 ml/min. The dissipated power density for both the ACHS and the MFHS was increased to ~ 100 W/cm². The total heated area is 0.6cm × 0.6cm.

The experimental results clearly indicate the performance gains offered by liquid cooling. Figure 5.21 shows the on-chip temperature rise as a function of localized heating power under various two DI water flow rates of 50ml/min and 75ml/min (heating area of ~0.36 cm²). As expected, when increasing the flow rate to 75ml/min, the temperature of the chip and thermal resistance of the microchannel heat sink decrease. However, as shown in the graph, the difference in chip temperature is not substantial. It can be assumed that the thermal resistance component R_{heat} , the resistance due to heating of the cooling fluid, which is disproportional to the fluid flow rate, is not the dominant thermal resistance component for the testing parameters previously described. It is likely that R_{conv} , the convective resistance, which is dependent on the heat transfer coefficient of

the cooling fluid and the area of the surfaces of the microchannels in contact with the cooling fluidic is the dominant thermal resistance component.

A summary of the experiential thermal measurements is shown in Table 5.3. The ACHS was shown to have a thermal resistance of 0.518°C/W at an average chip temperature of 77.6°C. The microchannel heat sink was shown to have a thermal resistance of 0.296°C/W at an average chip temperature of 49.9°C when cooling fluid is circulated through the microchannels at a flow rate of 50ml/min, and the microchannel heat sink is shown to have a thermal resistance of 0.27°C/W at an average chip temperature of 48.5°C when cooling fluid is circulated through the microchannels at a flow rate of 0.27°C/W at an average chip temperature of 48.5°C when cooling fluid is circulated through the microchannels at a flow rate of 75ml/min.



Figure 5.21: On-chip temperature rise as a function of localized heating power under various DI water flow rates (heating area of $\sim 0.36 \text{ cm}^2$).

	ACHS @ 54.8 CFM (~2500 rpm)	MCHS @ 50 ml/min	MCHS @ 75 ml/min
Thermal Resistance (°C·cm²/W)	0.518	0.296	0.270
Localized power density (W/cm ²)	109.6	105.7	105.9
Avg. Junction Temperature (°C)	77.6	49.91	48.54

Table 5.3: Comparison of heat sink thermal resistance and chip average junction temperature for air cooling vs. microchannel liquid cooling.

5.8. Hydraulic Analysis of Microchannel Heat Sink

While circulating fluid through the microchannels in the 3D chip stack, a differential pressure gauge was used to measure the pressure drop in a single microfluidic chip and the pressure drop in a two-layer 3D chip stack. Additionally, the pressure drop was measured at various fluidic flow rates in order to analyze the impact of the flow rate on the pressure drop in the microchannels. Figure 5.22 shows the results of the measured pressure drop in a single microfluidic chip as a function of fluidic flow rate, and Figure 5.23 shows the pressure drop in a two-layer 3D chip stack as a function of fluidic flow rate.

As expected, in both cases, the pressure drop increases as the fluidic flow rate increases. Furthermore, it is important to note that the average pressure drop across the 3D chip stack is approximately twice the amount of the average system pressure drop for a single chip. Consequently, the fluidic flow rate will have to be increased to supply a sufficient amount of fluid for every layer that is added to the 3D chip stack. Increasing

the fluidic flow rate for each additional layer ultimately means that the pressure drop of the 3D system increases for each additional layer in the 3D stack.



Figure 5.22: Pressure drop as a function of flow rate for a single microfluidic chip.



Figure 5.23: Pressure drop as a function of flow rate for a 3D stack of microfluidic chips (two layers).

5.9. Microfluidic Pin-fin Heat Sink

Although microchannel heat sink technology has been the primary focus of this research, alternative heat sink technologies were also investigated. An alternative heat sink technology to the microchannel heat sink is the pin-fin heat sink [5.1-5.4], which consists of an array of staggered circular pin-fin structures (Figure 5.24 and Figure 5.25) [5.5]. The 3D liquid cooling platform that has been developed enables the fabrication and thermal testing of alternative heat sink structures by changing one masking step in the process flow.

Preliminary fabrication and thermal analysis reveal that the pin-fin heat sink technology enables the fabrication of heat sink architecture with a much shorter height, while maintaining a similar thermal resistance as the microchannel heat sink. This result

is important because if electrical TSVs can be integrated in the pin-fin heat sink, electrical TSV fabrication will be easier due to the pin-fin heat sink being a shorter structure. Additionally, as previously discussed, shorter, high-aspect ratio electrical TSVs are preferred for signaling. Thinner and shorter TSVs result in faster interconnects, larger bandwidth density, and lower power consumption.

Table 5.4 summarizes the trade-offs between thermal resistance and pressure drop when comparing a microchannel heat sink to a pin-fin heat sink. The experiential results for the pin-fin heat sink are obtained from [5.5], and the experimental results for the microchannel heat sink are obtained from the most recent thermal measurements for microchannel heat sink analysis done in this work.



Figure 5.24: Top view of a SEM image of an array of micro pin-fins [5.5].



Figure 5.25: Top view SEM image of a micro pin-fin heat sink etched onto the back side of a silicon chip. Inlet and outlet channels that are used for routing cooling fluid to the pin-fin heat sink are shown on the left and right sides of the pin-fin heat sink [5.5].

Heat Sink Technology	R _{tot} (°C/W)	ΔP (kPa)	Flow Rate (ml/min)	Geometry
Pin-fin	0.269	28.23	70	D=150µm, Pitch=225µm, H _{fin} =200µm
Microchannel	0.27	18	75	Ww=Wc=80µm H _{ch} = 350µm

Table 5.4: Comparison of microchannel and micro pin-fin heat sinks [5.5].

A staggered pin-fin heat sink is shown to have a thermal resistance as low as 0.269°C/W, when circulating cooling liquid at a flow rate of 70 ml/min in a heat sink where the pin-fins have a height of 200µm. The microchannel heat sink is shown to have a thermal resistance of 0.27°C/W, when circulating cooling liquid at a flow rate of 75 ml/min in a heat sink where the microchannels have a height of 350µm. While the pressure drop in the microchannel heat sink is lower than that of the pin-fin heat sink, the height of the pin-fin heat sink is only 57% of the height of the microchannel heat sink. Consequently, ease of electrical TSV fabrication and improved electrical TSV

performance could be achieved by implementing a pin-fin heat sink as the cooling solution for thermal management of high-performance microprocessors in a 3D chip stack [5.5].

5.10. Summary

Chapter 5 describes the experimental thermal measurement test-bed for evaluating a 3D inter-layer liquid cooling platform that has been developed. Experimental thermal testing results for an air-cooled chip and a liquid-cooled chip are compared. The on-chip integrated microchannel heat sink is shown to have a thermal resistance of >0.28 °C/W and cooling of >100W/cm² high-power density chip at an average junction temperature of less than 50°C is demonstrated. Electrical and fluidic interconnection between layers is also demonstrated. Cooling liquid is circulated through the 3D stack at flow rates of up to 100 ml/min.

Preliminary fabrication results for an alternative pin-fin heat sink technology are demonstrated, and the advantages of the pin-fin heat sink compared to the microchannel heat sink technology are discussed.

CHAPTER 6

CONCLUSION AND FUTURE WORK

This chapter provides a summary of the key results presented in the dissertation. Additionally, ideas regarding how this work may be extended in future research efforts are discussed.

6.1. Opportunities for Future Work

1. High density C4 electrical I/Os and integrated fluidic I/Os

In this work, an electrical I/O density of 1600/cm² is demonstrated. However, according to ITRS projections, high-performance chips in a 3D stack may have up to 9000 die-to-die bonds by the year 2022 [1.11]. Consequently, high-density electrical I/O interconnections will be necessary. There is an opportunity for future work to demonstrate high density electrical I/O interconnections integrated with fluidic I/O interconnections.

2. Demonstration of silicon carrier for supplying liquid to multiple 3D chip stacks

This work demonstrates fluid being supplied to the microchannels in the 3D chip stack by inlet channels that are fabricated in the bottom layer of the 3D stack. According to the ITRS, as shown in Table 1.2, there could be up to 10 high-performance chips in a 3D package by the year 2022 [1.11]. Having this many chips in a package could require that there be multiple 3D chip stacks that contain high-performance processors. Consequently, if microfluidic cooling were implemented as the cooling solution, the

substrate on which the 3D ICs are stacked will be an important component of the 3D system. There is an opportunity for future experimental work in the implementation of a silicon carrier with integrated electrical and microfluidic interconnect networks. Within the silicon carrier, integrated fluidic interconnections can be developed to enable the routing of a coolant to multiple stacks of 3D ICs bonded on the silicon carrier.



3. Demonstration of polymer fluidic I/Os that supply liquid to multiple CPU chips in the 3D chip stack

As the design of 3D chip stacks that contain high-performance microprocessors becomes more complex, future 3D stacks have the potential to include multiple memory chips and multiple processor chips, such as the configuration shown in Figure 6.X.

In order to route fluid to high-performance processors in the 3D stack, a fluidic I/O technology that can traverse longer distances may be required (equivalent to the thickness of multiple memory chips). There is an opportunity to stack multiple memory and processor chips, while using polymer fluidic I/Os to supply liquid to multiple processor chips in the 3D stack.



4. Independent control of liquid coolant supplied to each layer of the 3D stack

In this work, fluidic is supplied to microchannels in the 3D chip stack by inlet channels that are fabricated in the bottom layer of the 3D stack. Because the inlet channels are in the bottom layer of the stack, fluid is easily supplied to the microchannels in the bottom layer of the stack. For fluid to be circulated to channels in each ascending layer, a sufficient fluid flow rate (and pumping power) must be applied to the microfluidic system in order for channels in the ascending layers in the stack to receive a sufficient amount of fluid for cooling. Alternative system configurations should be investigated to analyze the benefits of independent fluid flow to each level in the 3D stack.

6.2. Conclusion

Three-dimensional integration is a promising technology that offers many advantages including increased device density, shorter interconnect distances, system performance enhancements, decreased system form factor, and integration of heterogeneous technologies in the same chip stack. However, heat removal is a primary limiter of stacking high-performance microprocessors in a 3D chip stack. This research has demonstrated the design, fabrication, assembly, and experimental testing of a 3D integration platform that can support the heat removal requirements for high-performance chips.

6.2.1. Development of Inter-layer Liquid Cooling Platform for Thermal Management of 3D Integrated Circuits

Using microchannel inter-layer liquid cooling to facilitate cooling on each strata allows chips to be arranged in the 3D stack as needed to support the most efficient heat removal and power delivery methods. The proposed microchannel liquid cooling scheme can be used when stacking multiple processors, as this scheme significantly reduces the overall thermal resistance of the cooling system, removes thermal resistances associated with TIMs, reduces chip cooling hardware size from inches to microns, and enables cooling of >100W/cm² of each high-power density chip.

Back-end-of-the-line (BEOL) compatible, wafer-level batch fabrication and micromachining technologies are used to fabricate the necessary electrical and thermal interconnects for the proposed 3D platform. Each silicon die of the 3D stack contains a monolithically integrated microchannel heat sink, through-silicon fluidic vias used for fluidic routing in the 3D stack, solder bumps, and microscale fluidic I/Os on the side of the chip opposite to the microchannel heat sink. Fabrication results of the individual components and integration of the components is demonstrated. Fabrication processes for integrating microchannel heat sinks and electrical through-silicon vias are also demonstrated.

6.2.2. Compact Physical Modeling to Determine the Design Trade-offs for Microchannel Heat Sink and Electrical TSV Integration

In order to determine how to design the microchannel heat sink to meet future ITRS high-performance chip cooling requirements, compact physical modeling is used to analyze the impact of microchannel geometry and fluid flow rates on thermal resistance and pressure drop of the heat sink. Compact physical modeling is also used to explore the electrical TSV performance and microchannel heat sink cooling trade-offs when integrating microchannel heat sinks and electrical TSVs in a 3D chip stack.

Modeling results show that increased microchannel cooling capability requires increased wafer thickness. However, TSV fabrication becomes more difficult and TSV electrical performance decreases as wafer thickness increases. Furthermore, electrical TSV density increases as the width of microchannel wall increases, which decreases the number of microchannels that can be fabricated on the chip and thereby decreases cooling capability. Analyzing the trade-offs of microchannel heat sink cooling performance enables a better understanding of how to design a 3D system that contains integrated microchannel heat sinks and electrical TSVs.

6.2.3. Integrated Electrical and Fluidic I/O Technologies for 3D Inter-layer Cooling

In the demonstrated 3D cooling platform, microscale fluidic interconnection between strata is enabled by through-wafer fluidic vias and fluidic I/O interconnects. The fabrication and process integration techniques are outlined for three distinct fluidic I/O technologies including a C4 pipe fluidic I/O, an air-gap C4 fluidic I/O, and a polymer pipe fluidic I/O interconnect technology. Characterization of the electrical and fluidic I/O interconnect structures is performed, and an electrical I/O density of ~1600/cm² is demonstrated. The advantages and disadvantages of the three fluidic I/O technologies are discussed.

6.2.4. 3D Chip Stacking Using Electrical and Fluidic I/O Interconnects

Significant challenges in using microfluidic cooling in a 3D chip stack are the flip-chip bonding processes required to stack microfluidic chips and the ability to seal the fluidic interfaces in the 3D stack in order to prevent fluid leakage.

The ability to assemble chips with integrated electrical and fluidic I/Os and seal fluidic interconnections at each strata interface is demonstrated using three assembly and fluidic sealing techniques. Assembly results show the stacking of up to four chips that contain integrated electrical and fluidic I/O interconnects. Fluidic testing is performed by circulating fluid through the 3D stack at flow rates up to 100ml/min, with no fluid leakage observed.

6.2.5. Thermal and Fluidic Testing of the 3D Cooling Platform

Chapter 5 describes the experimental thermal measurement test-bed for evaluating the 3D inter-layer liquid cooling platform that has been developed. Experimental thermal testing results for an air-cooled chip and a liquid-cooled chip are compared. The on-chip integrated microchannel heat sink is shown to have a thermal resistance of >0.28 °C/W and cooling of >100W/cm² high-power density chip at an average junction temperature of less than 50°C is demonstrated. Microchannel heat sink cooling shows a significant junction temperature and heat sink thermal resistance reduction compared to air-cooling.

Preliminary fabrication results for a pin-fin heat sink technology are demonstrated, and the advantages of the pin-fin heat sink compared to the microchannel heat sink technology are discussed.

123

APPENDIX A

PUBLICATIONS BASED ON PRESENTED RESEARCH

Conference Proceedings:

- [1] C.R. King, J. Zaveri, H. S. Yang, M. S. Bakir, and J. D. Meindl, "Electrical and Fluidic C4 Interconnections for Inter-layer Liquid Cooling of 3D ICs," 60th Electronic Components and Technology Conference, Las Vegas, NV, June 2010.
- [2] C.R. King, J. Zaveri, H. S. Yang, M. S. Bakir, and J. D. Meindl, "Electro-fluidic C4 Interconnections for Inter-layer Liquid Cooling of 3D ICs," SRC TECHCON, Austin, TX, September 2009. *Received Packaging and Cooling "Best in Session" Paper Award*
- C.R. King, D. Sekar, M. Bakir, B. Dang, J. Pikarsky, J. Meindl, "3D Stacking of Chips with Electrical and Microfluidic Interconnects," in Proc. IEEE 58th Electronic Components and Technology Conference, Lake Buena Vista, FL, May 27-30, 2008.
 -Received ECTC Motorola Electronic Packaging Fellowship Outstanding Paper Award
- [4] C.R. King, D. Sekar, M. Bakir, B. Dang, J. Pikarsky, J. Meindl, "3D Chip Stacking with Inter-layer Liquid Cooling," SRC TECHCON, Austin, TX, November 2008.
- [5] C.R. King, D. Sekar, M. Bakir, B. Dang, J. Pikarsky, J. Meindl, "Assembly Techniques for Microfluidic Networks in Three-dimensional Integrated Circuits," SRC TECHCON, Austin, TX, September 10-12, 2007.
- [6] Y. Zhang, C. R. King, J. Zaveri, Y. J. Kim, V. Sahu, Y. Joshi, M. S. Bakir, "Coupled Electrical and Thermal 3D IC Centric Microfluidic Heat Sink Design and Technology," in Proc. IEEE 58th Electronic Components and Technology Conference, Lake Buena Vista, FL, May 31 - June 3, 2011.
- J. Zaveri, C. King Jr., H.S. Yang, M.S. Bakir, "Wafer-level Batch Fabrication of Silicon Microchannel Heat Sinks and Electrical Through Silicon Vias for 3D ICs," IMAPS 42nd International Symposium on Microelectronics, San Jose, CA, November 2009. -Received IMAPS Best Student Paper Award
- [8] J. Zaveri, C.R. King, H. S. Yang, and M. S. Bakir, "Wafer-level Batch Fabrication of Silicon Microchannel Heat Sinks and Electrical Through-silicon Vias," SRC TECHCON, Austin, TX, September 2009.
- [9] D. Sekar, C. King, B. Dang, M. Bakir, J. Meindl, "Removing heat from 3D stacked chips," Future Fab International, Vol. 29, vol. 4, pp. 80-85, April 2009. (*invited*)
- [10] D. Sekar, C. R. King, B. Dang, T. Spencer, H. Thacker, P. Joseph, M. Bakir and James Meindl," A 3D-IC Technology with Integrated Microchannel Cooling," IEEE International Interconnect Technology Conference, Burlingame, CA, June 1-4, 2008. -Received IITC Best Student Paper Award
- [11] M. S. Bakir, C. King, D. Sekar, H. Thacker, B. Dang, G. Huang, A. Naeemi, and J. D. Meindl, "3D heterogeneous integrated systems: liquid cooling, power delivery, and implementation," in Proc. IEEE Custom Integrated Circuits Conference, 2008.
- [12] M. S. Bakir, C. King, D. Sekar, and B. Dang, "Electrical, Optical, Thermal Interconnect Networks for 3D Heterogenous Integrated Systems," in Proc. IEEE Avionics, Fiber-Optics and Photonics Conference, 2008.
- [13] M. S. Bakir, P. A. Thadesar, C. King, J. Zaveri, H. S. Yang, C. Zhang, Y. Zhang, "Revolutionary Innovation in System Interconnection: New Era for the IC," in Proceedings of SPIE Photonics West, January 2011.
- [14] T. Shodeinde, M.S. Bakir, C. R. King, "Sidewall Metallization of High Aspect Ratio Perpendicular Polymer Structures for Chip I/O Interconnections," National Nanotechnology Infrastructure Network (NNIN) REU Convocation, Cornell University, Ithaca, NY, August 2006.

Journal Publications:

- [1] B. Dang, M. Bakir, D. Sekar, C. King, and J. Meindl, "Single and 3D chip cooling using microchannels and microfluidic chip input/output (I/O) interconnects," IEEE Trans. Adv. Packaging, January 2010.
- [2] M. Bakir, G. Huang, D. Sekar, and C. King, "3D system integration: power delivery, cooling, and signaling," IETE Technical Review, November 2009. (invited)

[3] J. Lai, H. Yang, H. Chen, C. King, J. Zaveri, R. Ravindran, and M. Bakir, "Improved through-silicon-via fabrication for thick wafers containing MEMS and sensor devices," J. of Microelectromechanics & Microengineering, 2010.

Patents:

- [1] C. R. King, J. Zaveri, M. S. Bakir, and J. D. Meindl, "Air-Gap C4 Fluidic I/O Interconnects and Methods of Fabricating Same," filed March 2012, Serial No.: 13/416,948, patent pending.
- [2] M. S. Bakir, D. Sekar, C.R. King, B. Dang, J. Meindl, "3D ICs with Microfluidic Interconnects," issued April 19, 2011, patent number 7,928,563.

Book Chapter:

[1] B. Dang, M. Bakir, D. Sekar, C. King, and J. Meindl, "Single and 3D Chip Cooling using Microchannels and Fluidic I/Os," in *Integrated Interconnect Technologies* for 3D Nanoelectronic Systems, M. Bakir and J. Meindl (Eds.), Artech House, 2008. (invited)

REFERENCES

Chapter 1: Introduction and Background

- [1.1] J. Bardeen and W. H. Brattain, "The Transistor, A Semiconductor Triode," Phys. Rev., 74, 230-231 (1948).
- [1.2] J. Kilby, U.S. Patent 3,138,743, *Miniaturized Electronic Circuits*, filed February 6, 1959, issued June, 1964.
- [1.3] R. Noyce, U.S. Patent 2,981,877, Semiconductor Device and Lead Structure, filed July 1959, issued April 1961, assigned to Fairchild Semiconductor.
- [1.4]Texas Instruments, *TI People: Jack Kilby*, Retrieved from http://www.ti.com/corp/docs/kilbyctr/downloadphotos.shtml, (Accessed December 15, 2010).
- [1.5] G. E. Moore, "Cramming more components on the integrated circuits", *Electronics*, vol. 38, no. 8, 19 April 1965.
- [1.6] D. J. Paul, University of Cambridge, *Moore's Law*, Retrieved from http://www.sp.phy.cam.ac.uk/~SiGe/Moore%27s%20Law.html, (Accessed December 17, 2010).
- [1.7] M. Pedram and S. Nazarian, "Thermal Modeling, Analysis, and Management in VLSI Circuits: Principles and Methods," *in Proceedings of the IEEE*, Vol. 94, No. 8, August 2006.
- [1.8] S. Lin and K. Banerjee, "Cool Chips: Opportunities and Implications for Power and Thermal Management," in *IEEE Transactions on Electronic Devices*, Vol. 55, No. 1, January 2008.
- [1.9] C. Isci, PhD dissertation, "Workload Adaptive Power Management with Live Phase Monitoring and Prediction, Department of Electrical Engineering, Princeton University, 2007.

- [1.10] M. S. Bakir, B. Dang, and J. D. Meindl, "Revolutionary nanosilicon ancillary technologies for ultimate-performance gigascale systems," *IEEE Custom Integrated Circuits Conference*, 2007.
- [1.11]International Technology Roadmap for Semiconductors, 2010 Edition, Semiconductor Industry Association, 4300 Stevens Suite Boulevard, Suite 271, San Jose, CA 95129.
- [1.12] Z. J. Zhou, L. R. Hoover, and A. L. Philips, "An integrated thermal architecture for thermal management of high power electronics," in *Proc. of the International Conference on THERMES*, pp. 317-336, 2002.
- [1.13] M. S. Bakir and J. D. Meindl, Integrated Interconnect Technologies for 3D Nanoelectronic Systems, Artech House, 2009.
- [1.14] G. Shahidi, "Evolution of CMOS Technology at 32 nm and Beyond," *IEEE Custom Integrated Circuits Conference*, 2007.
- [1.15]J. Parkhurst, J. Darringer, and B. Grundmann, "From Single Core to Multi-Core: Preparing for a new exponential," *in Proc. IEEE/ACM International Conference on Computer-Aided Design*, pp. 67-72, 2006.
- [1.16] M. B. Healy, H. S. Lee, G. H. Loh, and S. K. Lim, "Thermal Optimization in Multi-Granularity Multi-Core Floorplanning," in Proceedings Design Automation Conference, 2009.
- [1.17]P.G. Emma and E. Kursun. "Is 3D chip technology the next growth engine for performance improvement?," *IBM Journal of Research and Development*, Vol. 52, pp. 541-52, November 2008.
- [1.18] J. D. Meindl, "Beyond Moore's Law: the interconnect era," IEEE Computing Science and Engineering, vol. 5, no. 1, pp. 20-24, Jan/Feb 2003.
- [1.19] J. D. Meindl, "Low power microelectronics: Retrospect and prospect," Proc. IEEE, vol. 83, no. 4, pp. 619–635, Apr. 1995.
- [1.20] J. R. Powell, "The Quantum Limits to Moore's Law," in Proceedings of the IEEE, Vol. 96, No. 8, August 2008.

- [1.21] D. Martell, "Intel's Moore muses on end of technology maxim," *Reuters*, Sep. 19, 2007.
- [1.22] V. V. Zhirnov, R. K. Calvin, J. A. Hutchby, and G. I. Bourianoff, "Limits to binary logic switch scaling—A Gedankan model," *Proc. IEEE*, vol. 91, no. 11, pp. 1934– 1939, Nov. 2003.
- [1.23] L. B. Kish, "End of Moore's law: Thermal (noise) death of integration in micro and nanoelectronics," *Phys. Lett. A*, vol. 305, pp. 144–149, 2002.
- [1.24] D. J. Frank, R. H. Dennard, E. Nowak, P. M. Solomon, Y. Taur, and H. P. Wong, "Device scaling limits of Si MOSFETs and their application dependencies," *Proc. IEEE*, vol. 89, no. 3, pp. 259–288, Mar. 2001.
- [1.25] S. Krishnan, S. V. Garimella, G. M. Chrysler, and R. V. Mahajan, "Towards a Thermal Moore's Law," *IEEE Transactions on Advanced Packaging*, Vol. 30, No. 4, August 2007.
- [1.26] A. W. Topol, D. C. La Tulipe, Jr., L. Shi, D. J. Frank, K. Bernstein, S. E. Steen, A. Kumar, G. U. Singco, A. M. Young, K. W. Guarini, M. Ieong, "ThreeDimensional Integrated Circuits", *IBM Journal of Research &Development*, Vol. 50, Number 4/5, 2006.
- [1.27] J. U. Knickerbocker, P. S. Andry, B. Dang, R. R. Horton, M. J. Interrante, C. S. Patel, R. J. Polastre, K. Sakuma, R. Sirdeshmukh, E. J. Sprogis, S. M. Sri Jayantha, A. M. Stephens, A. W. Topol, C. K. Tsang, B. C. Webb, S. L. Wright, "Three-dimensional Silicon Integration," *IBM Journal of Research & Development*, Vol. 52, Number 6, 2008.
- [1.28] K. Takahaski, Y. Taguchi, M. Tomisaka, H. Yonemara, M. Hoshino, M. Ueno, Y. Egawa, et al., "Process Integration of 3D Chip Stack with Vertical Interconnection," *Proceedings of the 54th Electronic Components and Technology Conference*, June 1–4, 2004, pp. 601–609.
- [1.29] B. Dang, S. L. Wright, P. S. Andry, E. J. Sprogis, C. K. Tsang, M. J. Interrante, B. C. Webb, R. J. Polastre, R. R. Horton, C. S. Patel, A. Sharma, J. Zheng, K. Sakuma, and J. U. Knickerbocker, "3D chip stacking with C4 technology," *IBM J. Res. Develop.*, vol. 52, no. 6, pp. 599–609, Nov. 2008.

- [1.30] J. U. Knickerbocker, C. S. Patel, P. S. Andry, C. K. Tsang, L. P. Buchwalter, E. J. Sprogis, H. Gan, et al., "3-D Silicon Integration and Silicon Packaging Technology Using Silicon Through-Vias," IEEE J. Solid-State Circuits 41, No. 8, 1718–1725, 2006.
- [1.31] M. Karnezos, "3-D Packaging: Where All Technologies Come Together," in Proceedings of IEEE/CPMT/SEMI 29th International Electronics Manufacturing Technology Symposium, 2004.
- [1.32] H. Ikeda, M. Kawano, and T. Mitsuhashi, "Stacked Memory Chip Technology Development," *in Proceedings SEMI Technology Symposium (STS)*, 2005.
- [1.33] B. Black, D. W. Nelson, C. Webb, and N. Samra, "3D processing technology and its impact on iA32 microprocessors," *in Proc. IEEE Int. Conf. Computer Design*, pp. 316–318, 2004.
- [1.34] M. S. Bakir, C. King, D. Sekar, H. Thacker, B. Dang, G. Huang, A. Naeemi, and J. D. Meindl, "3D heterogeneous integrated systems: liquid cooling, power delivery, and implementation," *in Proc. IEEE Custom Integrated Circuits Conference*, 2008.
- [1.35] John H. Lau and Tang Gong Yue, "Thermal Management of 3D IC Integration with TSV (Through Silicon Via)," in Proc. IEEE 59th Electronic Components and Technology Conference, 2009.
- [1.36] A. Jain, R. Jones, R. Chatterjee, S. Pozder, Z. Huang, "Thermal Modeling and Design of 3D Integrated Circuits," 11th Inter-society Conference on Thermal and Thermomechanical Phenomena in Electronic Systems, 2008.
- [1.37] J. Wei, "Thermal Management of Fujitsu's High-performance Servers," *Fujitsu Science and Technology Journal*, 43, 1, pp. 122-129, January 2007.
- [1.38] C. M. Tan and F. He, "3D circuit model for 3D IC reliability study," in Proc. 10th International Conference on Thermal, Mechanical and Multi-Physics simulation and Experiments in Microelectronics and Microsystems, pp. 1-7, 2009.
- [1.39] D. B. Tuckerman and R. F. W. Pease, "High performance heat sinking for VLSI," IEEE Electron Device Letter, vol. EDL-2, pp.126-129, 1981.

- [1.40] S. P. Gurrum, S. K. Suman, Y. K. Joshi, and A.G. Fedorov, "Thermal issues in next-generation integrated circuits," *IEEE Transactions on Device and Materials Reliability*, vol. 4, pp. 709-14, 2004.
- [1.41] J. Koo, S. Im, and L. Jiang, "Integrated microchannel cooling for three dimensional electronic circuit architectures," *Journal of Heat Transfer*, Volume 127, January 2005.
- [1.42] R. Viswanath, V. Wakharkar, A. Watwe, V. Lebonheur, "Thermal performance challenges from silicon to systems," *Intel Technology Journal*, vol. 8 no. 4, pp.1-16, 2000.
- [1.43] T. Thompson. "Exploring options for keeping the heat out." *Chipscale Review: Thermal Management*, August/September 2006.

<u>Chapter 2: Development of an Inter-layer Liquid Cooling Platform for Thermal</u> <u>Management of 3D Integrated Circuits</u>

- [2.1] B. Dang, M. Bakir, D. Sekar, C. King, and J. Meindl, "Single and 3D chip cooling using microchannels and microfluidic chip input/output (I/O) interconnects," IEEE Trans. Adv. Packaging, January 2010.
- [2.2] N. H. Khan, S. M. Alam, and S. Hassoun, "System-Level Comparison of Power Delivery Design for 2D and 3D ICs," in Proceedings of 3DIC IEEE International Conference on 3D System Integration, 2009.
- [2.3] C.R. King, D. Sekar, M. Bakir, B. Dang, J. Pikarsky, J. Meindl, "3D Stacking of Chips with Electrical and Microfluidic Interconnects," in Proc. IEEE 58th Electronic Components and Technology Conference, 2008.
- [2.4] B. Dang, M. Bakir, J. Meindl, "Integrated thermal-fluidic I/O interconnects for an on-chip microchannel heat sink," *IEEE Electron Device Letters*, vol. 27, no. 2, pp. 117-119, 2006.
- [2.5] J. Zaveri, C. King Jr., H.S. Yang, M.S. Bakir, "Wafer level batch fabrication of silicon microchannel heat sinks and electrical through silicon vias for 3D ICs," *IMAPS 42nd International Symposium on Microelectronics*, San Jose, CA, November 2009.

- [2.6] J. Lai, H. Yang, H. Chen, C. King, J. Zaveri, R. Ravindran, and M. Bakir, "Improved through-silicon-via fabrication for thick wafers containing MEMS and sensor devices," Journal of Microelectromechanics & Microengineering, 2010.
- [2.7] D. Sekar, C. R. King, B. Dang, T. Spencer, H. Thacker, P. Joseph, M. Bakir and James Meindl, "A 3D-IC Technology with Integrated Microchannel Cooling," *in Proc. IEEE International Interconnect Technology Conference*, Burlingame, CA, June 1-4, 2008.
- [2.8] B. Dang, *Integrated Input/Output Interconnection and Packaging for GSI*, Ph.D. Dissertation, Georgia Institute of Technology, 2006.
- [2.9] C.R. King, J. Zaveri, H. S. Yang, M. S. Bakir, and J. D. Meindl, "Electrical and Fluidic C4 Interconnections for Inter-layer Liquid Cooling of 3D ICs," 60th ECTC, Las Vegas, NV, June 2010.
- [2.10]Dragoi, V., Mittendorfer, G., Thanner, C., and Lindner, P., \Wafer-level plasma activated bonding: new technology for mems fabrication," Microsystem Technologies, vol. 14, pp. 509-515, 2008. 10.1007/s00542-007-0437-7.
- [2.11] Reiche, M. and Wiegand, M., Plasma-enhanced semiconductor wafer bonding: Mechanism and applications," in: Semiconductor Wafer Bonding: Science, Technology, and Applications V. (Eds: Hunt, C.E., Baumgart, H. and Goesele. U.), The Electrochemical Society, Pennington, PV 99-35 (1999).
- [2.12]Farrens, S. N., Dekker, J. R., Smith, J. K., and Roberds, B. E., Chemical free room temperature wafer to wafer direct bonding," Journal of the Electrochemical Society, vol. 142, no. 11, pp. 3949-3955, 1995.
- [2.13] Suni, T., Henttinen, K., Suni, I., and Makinen, J., "Effects of plasma activation on hydrophilic bonding of Si and SiO₂," Journal of The Electrochemical Society, vol. 149, no. 6, pp. G348-G351, 2002.
- [2.14] D. Liu and S. V. Garimella, "Analysis and optimization of the thermal performance of microchannel heat sinks," *International Journal of Numerical Methods for Heat & Fluid Flow*, vol. 15, pp. 7-26, 2005.
- [2.15]P. Lee, J. Ho, and H. Xue, "Experimental study on laminar heat transfer in microchannel heat sink," in *Proc. Inter society Conference on Thermal Phenomena* (*IEEE ITherm*), pp.379-386, 2002.

- [2.16]Shah, R.K. and London, A.L., Laminar Flow Forced Convection in Ducts, Academic Press, New York, 1978.
- [2.17]M. S. Bakir, C. King, D. Sekar, H. Thacker, D. Bing, H. Gang, A. Naeemi, and J. D. Meindl, "3D heterogeneous integrated systems: Liquid cooling, power delivery, and implementation," in *Proc. Custom Integrated Circuits Conference*, 2008, pp. 663-670.
- [2.18]G. Katti, M. Stucchi, K. De Meyer, and W. Dehaene, "Electrical Modeling and Characterization of Through Silicon via for Three-Dimensional ICs," *IEEE Transactions on Electron Devices*, vol. 57, Jan. 2010, pp. 256-262.
- [2.19]Narain Arora *et al*, <u>Mosfet modeling for VLSI simulation: theory and practice</u>, World Scientific Publishing Company (2007), pp. 121-165.

Chapter 3: Design and Fabrication of Electrical and Fluidic I/O Interconnects

- [3.1] F. Schindler-Saefkow. O. Wittler, D. May, B. Michel Thermal Management in a 3D-PCB-Package with Water Cooling, *in Proc. Electronic Systems Integration Technology Conference*, pp.107-110, 2006.
- [3.2] D. Sekar, *Optimal Signal, Power, Clock and Thermal Interconnect Networks for High-Performance 2D and 3D Integrated Circuits*, Ph.D. Dissertation Georgia Institute of Technology, 2008.
- [3.3] M. S. Bakir, P. A. Thadesar, C. King, J. Zaveri, H. S. Yang, C. Zhang, Y. Zhang, "Revolutionary Innovation in System Interconnection: New Era for the IC," in Proceedings of SPIE Photonics West, January 2011.

<u>Chapter 5: Thermal and Fluidic Testing of the 3D Inter-layer Liquid Cooling</u> <u>Platform</u>

- [5.1] S. Ndao, Y. Peles, and M.K. Jensen, "Multi-objective thermal design optimization and comparative analysis of electronics cooling technologies," *International Journal of Heat and Mass Transfer*, vol. 52, Sep. 2009, pp. 4317-4326.
- [[5.2] D.E. Metzger, R.A. Barry, J.P. Bronson, Developing heat transfer in rectangular ducts with staggered arrays of short pin-fins, ASME Trans. J. Heat Transfer 104 (1982) 700–706.

- [5.3] S.Y. Won, G.I. Mahmood, P.M. Ligrani, Spatially-resolved heat transfer and flow structure in a rectangular channel with pin fins, Int. J. Heat Mass Transfer 47 (2004) 1731–1743.
- [5.4] A. Kosar, C. Mishra, Y. Peles, Laminar flow across a bank of low aspect ratio micro pin fins, J. Fluids Eng. 127 (2005) 419–430.
- [5.5] Y. Zhang, C. R. King, J. Zaveri, K. Yoon Jo, V. Sahu, Y. Joshi, and M. S. Bakir, "Coupled electrical and thermal 3D IC centric microfluidic heat sink design and technology," in *Proc. Electronic Components and Technology Conference*, 2011, pp. 2037-2044.

VITA

Calvin R. King, Jr. was born in Marianna, Arkansas in 1983. He received the Bachelor of Science in Electrical Engineering degree (summa cum laude) from Tennessee State University (TSU), Nashville, TN in 2005 and the Master of Science in Electrical and Computer Engineering and Master of Business Administration (MBA) degrees from the Georgia Institute of Technology (Georgia Tech) in 2007 and 2009, respectively. His research areas of interest include three-dimensional (3D) system integration, packaging and cooling for 3D systems, and micro-fabrication technology.

Awards received while at TSU include the TSU Presidential Scholarship, Coca-Cola National Scholarship, and TSU Department of Electrical and Computer Engineering Most Outstanding Senior. During his graduate studies, King has also received the IBM/ SRC Ph.D. Fellowship, IBM FOCUS Fellowship, Ford Foundation Fellowship, Applied Materials Ph.D. Fellowship, Motorola Electronic Packaging Fellowship, NASA Harriet Jenkins Pre-doctoral Fellowship, Georgia Tech FACES Fellowship, Qualcomm "Q" Award of Excellence, Intel Ph.D. Fellowship, and the Georgia Tech Tower Award.

He is the author/co-author of 17 conference proceedings and journal publications, 1 book chapter, and 2 US patents. He has been awarded the SRC Inventor Recognition Award (2008), the Motorola Electronic Packaging Fellowship Outstanding Student Paper Award from the 2008 *Electronic Components and Technology Conference* (ECTC), and the SRC TECHCON 2009 Packaging and Cooling Best in Session Paper Award.

King has worked with the IBM Systems and Technology group in Austin, TX and East Fishkill, NY, and he is a member of the National Society of Black Engineers (NSBE) and Institute of Electrical and Electronics Engineers (IEEE).